IMPACT OF PETROLEUM RELATED COMPOUNDS ON MESENCHYMAL STEM CELL DERIVED PROGENITOR CELLS

IMPACT OF PETROLEUM RELATED COMPOUNDS ON MESENCHYMAL STEM CELL DERIVED PROGENITOR CELLS

By ROBERT MICHAEL GUTGESELL, H.BSc.

A Thesis Submitted to the School of Graduate Studies in Partial Fulfillment of the Requirements for the Degree Master of Science

McMaster University © Copyright by Robert Michael Gutgesell, February 2022

McMaster University MASTER OF SCIENCE (2021) Hamilton, Ontario (Medical Sciences)

TITLE: IMPACT OF PETROLEUM RELATED COMPOUNDS ON MESENCHYMAL STEM CELL DERIVED PROGENITOR CELLS AUTHOR: Robert Michael Gutgesell, H.BSc. (McMaster University) SUPERVISOR: Professor Alison C. Holloway, PhD NUMBER OF PAGES: xii, 118

Lay Abstract

There is concern that chemicals from oil sands mining in the Athabasca oil sands region are hurting the reproductive health of animals in the wild. Some of these animals, including bears, wolves, and river otters, need a bone in their penis called a baculum to reproduce. Studies have shown that some chemicals, including those from mining activity can make the baculum bone weaker. For bone to stay strong, bone cells always need to be developing to fix the bone tissue. The goal of our study was to find how chemicals from mining activity can affect the development of bone cells. We found that a group of chemicals that come from oil sands mining called naphthenic acid fraction components (NAFCs) stop bone cells from developing and making new bone. We also know that having more fat cells in bone is associated with weaker bones. We also looked at whether NAFCs could increase the development of fat cells. However, NAFCs did not increase the development of fat cells. Together, this research shows that NAFCs can make bones like the baculum weaker by slowing the development of new bone, but not by increasing fat cells. Our research suggests that exposure to NAFCs may make baculums weaker which may be bad for the reproductive health of animals living near oil sands mining activity.

Abstract

There is concern over the impact that petroleum related compounds (PRCs) associated with mining activity in the Athabasca Oil Sands Region (AOSR) are having on local wildlife. With the increase in oil sands mining activity in the AOSR there has been a corresponding decline in the fertility of indicator species in the AOSR. One of the primary sources of PRCs in the environment is oil sands process affected water (OSPW), which is stored in tailings ponds. Several PRCs, including naphthenic acid fraction components (NAFC), have endocrine disrupting effects, which may, in part, explain reduced fertility in indicator species. For example, male North American river otters (Lontra canadensis) living in areas impacted by mining activity have lower baculum strength those unaffected by mining activity. Weaker baculums are associated with increases in fracture rates and reduced fertility. Baculum strength is maintained throughout life by bone remodeling, a process that requires the differentiation of osteoblasts. NAFCs can impact several pathways integral to the development and path selection of mesenchymal stem cells into osteoblasts or adjocytes. Therefore, the objective of this thesis was to test the hypothesis that NAFCs inhibit osteoblast differentiation and induce adipocyte differentiation from progenitor cells. We exposed osteoblast progenitor cells and adipocyte progenitor cells to NAFCs. We demonstrated that NAFCs inhibit osteoblast differentiation and activate the glucocorticoid receptor pathway. We also found that NAFCs do not induce adipogenesis in adipocyte precursor cells. Lastly, we

showed that NAFCs are PPAR γ ligands that inhibit the expression of PPAR γ associated genes. These insights into the effects of NAFCs on osteoblast and adipocyte progenitor cells suggest NAFCs may contribute to lower baculum strength and impaired adipose tissue function of animals living in the AOSR. These effects my reduce the fertility and population of wildlife in the AOSR.

Publications during the acquisition of this degree

Gutgesell, R. M., Tsakiridis, E. E., Jamshed, S., Steinberg, G. R., & Holloway, A.
C. (2020). Impact of pesticide exposure on adipose tissue development and function. *Biochemical Journal*, 477(14), 2639–2653. https://doi.org/10.1042/BCJ20200324

Acknowledgements

First, I would like to thank my supervisor, Dr. Alison Holloway for her mentorship, guidance, and support over the course of my MSc. degree. I am very appreciative of the opportunity to study in your lab. Thank you as well to my supervisory committee members, Dr. Karen Beattie and Dr. Cheryl Quenneville for their time and guidance in my research. I would also like to thank the current and former members of the Holloway lab, who have been good peers, friends, and trainees. To my partner, Selena, thank you for your patience, compassion, and brightening the way. Lastly, I would like to thank my family and friends, I am extremely grateful for your support, patience, and encouragement throughout this journey.

Table of Contents

LAY ABSTRACT	III
ABSTRACT	IV
PUBLICATIONS DURING THE ACQUISITION OF THIS DEGREE	VI
ACKNOWLEDGEMENTS	VII
TABLE OF CONTENTS	VIII
LIST OF FIGURES AND TABLES	IX
LIST OF ALL ABBREVIATIONS	X
DECLARATION OF ACADEMIC ACHIEVEMENT	XII
1. CHAPTER 1: INTRODUCTION	1
1.1.1. Naphthenic Acid Fraction Components 1.1.2. Mesenchymal stem cells 1.1.3 Bong	4 6 7
1.1.5. Bone marrow adipose tissue 1.1.4. Bone marrow adipose tissue 1.1.5. Pathways regulating MSC differentiation 1.2. MAIN OBJECTIVE	
2. CHAPTER 2	
 2.1. PREFACE AND SIGNIFICANCE TO THESIS 2.2. AUTHOR CONTRIBUTION 2.3. NAPHTHENIC ACID FRACTION COMPONENTS FROM OIL SANDS PROCESS AFFEC 	
WATER FROM THE ATHABASCA OIL SANDS REGION IMPAIR MURINE OSTEOBLAST DIFFERENTIATION AND FUNCTION	
3. CHAPTER 3	
 3.1. PREFACE AND SIGNIFICANCE TO THESIS	55 55 TED L1 CELLS
4. CHAPTER 4: DISCUSSION	88
4.1. Limitations and Future directions	
BIBLIOGRAPHY	

List of Figures and Tables

Chapter 1

Figure 1 - Comparison of measured fertility rate of otters in the AOSR and other locations in North America.

Chapter 2

- Figure 1 Effects of NAFC on cell viability.
- Figure 2 NAFC inhibits osteoblast differentiation in MC3T3-E1 cells in a dose dependent manner.
- Figure 3 Effects of NAFC on osteoblast markers during differentiation of
- MC3T3-E1 cells

Figure 4 - NAFC induce glucocorticoid receptor activity in MC3T3-E1 cells.

- Table 1 Table of primers.
- Table 2 Table of antibodies.

Chapter 3

Figure 1 - NAs contain PPARy ligands

Figure 2 - Effect of NAFC on adipogenesis and cell survival.

Figure 3 - RNA markers of PPARy induced adipogenesis NAFC exposed

adipocytes

Figure 4 - Lipid accumulation and markers of adipogenesis in mature adipocytes treated for 48 hours with NAFC

Table 1 – Table of primers.

List of all Abbreviations

ALP	Alkaline Phosphatase
aMEM	Alpha Minimum Essential Medium
AOSR	Athabasca Oil Sands Region
AP-1	Activator Protein 1
BCA	Bicinchoninic Acid Assay
BM	Bone Marrow
BMAT	Bone Marrow Adipose Tissue
BMD	Bone Mineral Density
BMP	Bone Morphogenic Protein
BSA	Bovine Serum Albumin
BTEX	Benzene, Toluene Ethylbenzene and Xylene
C/EBP	CCAAT/Enhancer Binding Protein
C/EBPβ	CCAAT Enhancer Binding Protein Beta
Cd36	Cluster of Differentiation 36
CNA	Commercial Naphthenic Acid
Col1a1	Collagen Type 1 alpha 1
CTL	Control
DMEM	Dulbecco's Modified Eagle Medium
ECL	Electrochemiluminescence
ECM	Extracellular Matrix
EDC	Endocrine Disrupting Chemical
EDTA	Ethylenediaminetetraacetic acid
FABP4	Fatty Acid Binding Protein 4
GR	Glucocorticoid Receptor
HEPES	4-(2-hydroxyethyl)-1-piperazineethanesulfonic acid
IBMX	3-isobutyl-1-methylxanthine
IGF-1	Insulin-like Growth Factor 1
IRF-3	Interferon Regulatory Transcription Factor 3
MSC	Mesenchymal Stem Cell
NA	Naphthenic Acid
NAFC	Naphthenic Acid Fraction Component
NF-kB	Nuclear Factor Kappa-Light-Chain-Enhancer of Activated B Cells
OPG	Osteoprotegerin
OSPW	Oil Sands Process Affected Water

p85a	Phosphatidylinositol 3-kinase Regulatory Subunit Alpha	
PAC	Polycyclic Aromatic Compound	
PPARγ	Peroxisome Proliferator-Activated Receptor Gamma	
PPIA	Peptidylprolyl Isomerase A	
PRC	Petroleum Related Compound	
Pref-1	Preadipocyte Factor-1	
PTH	Parathyroid Hormone	
PVDF	Polyvinylidene Difluoride	
RUNX2	Runt Related Transcription Factor 2	
SGK1	Serum/Glucocorticoid Regulated Kinase 1	
STAT5	Signal Transducer and Activator of Transcription 5	
Wnt	Wingless/integrated	
Ywhaz	Tyrosine 3-monooxygenase/tryptophan 5-monooxygnase Activation Protein Zeta	

Declaration of Academic Achievement

Robert Gutgesell (RMG) wrote the first draft of the thesis while Alison Holloway (ACH), Cheryl Quenneville, and Karen Beattie provided edits and feedback. In Chapter 1, ACH, Philippe Thomas (PJT), Jon Cormier and Sergio Raez-Villanueva were responsible for generating the data presented in Figure 1. In Chapter 2 RMG was the primary contributor to the data in Figure 1, Figure 2, Figure 3, and Figure 4. Laiba Jamshed (LJ) helped with some of the preliminary lab experiments. RMG wrote the first draft of the manuscript and the rest of the authors provided edits and feedback on the manuscript.

In Chapter 3, RMG was the primary contributor to the data in Figure 2, Figure 3, and Figure 4. LJ was the primary contributor to Figure 1. LJ and Raji Rajilingham helped with some of the preliminary lab experiments and cell culture. RMG wrote the first draft of the manuscript and the rest of the authors provided edits and feedback on the manuscript. For both Chapters 2 and 3 RMG, PJT and ACH were responsible for experimental design, and Mark Hewitt and Richard Frank provided the NAFC stock solution.

1. Chapter 1: Introduction

The Athabasca Oil Sands Region (AOSR) is one of the world's largest deposits of extractable bitumen. It is forecasted to contribute 1.7 trillion dollars to Canada's economy over the next decade (Doluweera et al., 2017). The extraction of bitumen requires a large amount of steam, producing 3-4 barrels of waste-water for every barrel of bitumen extracted (Canada, 2016). There are approximately 1.2 trillion litres of this wastewater stored in tailings ponds in the AOSR that needs to be isolated from natural ecosystems because of its contaminant load (Alberta Energy Regulator, 2019). There are four major categories of petroleum related contaminants (PRCs) in OSPW; naphthenic acids (NAs), polycyclic aromatic compounds (PACs), trace metals and BTEX (comprised of benzene, toluene ethylbenzene and xylene) (Huang et al., 2016). The primary routes of exposure for PRCs to the wildlife in the AOSR are seepage of oil-sands process affected water (OSPW) into natural watercourses, deposition into snowpack and atmospheric exposure to emissions from upgrading and other on-site industrial processes, and exposure to tailings ponds themselves (Fernie et al., 2018).



Figure 1. Comparison of measured fertility rate of otters in the AOSR and other locations in North America. Data for historical comparisons from **1**:(Government of New Brunswick, 2016), **2**:(Chilelli et al., 1996), **3**:(Barding EE, 2014), **4**:(Cumberland & DeVINK, 2017), **5**:(Stenson, 1975).

There is concern over the potential impact of PRCs on the local human and wildlife populations, and several PRCs have been associated with reduced fertility in mammals (Kassotis et al., 2016; Nagel et al., 2020; Sirotkin & Harrath, 2017). There is also evidence suggesting a decline in the reproductive rate of river otters (Lontra canadensis) in the AOSR relative to historical data from other North American populations (Fig 1.)(Raez-Villanueva et al., 2019). North American river otters are a member of the Mustelidae family and are sensitive indicator species due to their high trophic position and semi-aquatic habitat that makes them susceptible to contaminants in the rivers and lakes in the AOSR in addition to soil and air-based contaminants. Thomas et al. (2021) investigated whether proximity to oil sands mining activity in the AOSR influenced the baculum strength of river otters (Lontra canadensis). The baculum is a mineralized long bone found in the glans penis in many mammal species, and while its function is not entirely understood, it is thought to provide structural support during mating (Brassey et al., 2018). Thomas et al. found that river otters trapped in areas with high levels of mining activity had baculums that were less stiff and failed at a lower peak load than those trapped in areas within the AOSR that were unaffected by mining activity (Thomas et al., 2021). Bone strength can be affected by a variety of endocrine factors, including sex hormones and glucocorticoids that impact different parts of remodeling (Canalis & Delany, 2002; Khosla et al., 2012; Mohamad et al., 2016). Exposure to endocrine disrupting chemicals (EDC) has been shown to be associated with lower baculum bone mineral density (BMD) and increased likelihood of baculum fracture; effects that may impair reproductive success (Sonne et al., 2015). Thus, the decreased baculum strength observed by Thomas et al. may be a contributing factor in the low fertility rate observed in the river otters in the AOSR (Raez-Villanueva et al.,

2019). Several PRCs have endocrine disrupting effects, including some PACs that have been found to have steroidogenic effects (S. Lee et al., 2017), and several metals that have been reported to have estrogenic effects (Paschoalini et al., 2019; Yang et al., 2015). In addition, NAs extracted from OSPW can act as androgen and estrogen receptor antagonists (Leclair et al., 2015). Androgen and estrogen are involved in regulating bone remodeling by acting on both osteoblastogenesis and osteoclastosgenesis, and are required to maintain bone mass (Vanderschueren et al., 2004). Therefore, it is plausible that PRCs are contributing to the lower bone strength phenotype observed in animals living in close proximity to active mining sites in the AOSR.

1.1.1. Naphthenic Acid Fraction Components

NAs are present in all crude oil and are concentrated in OSPW as a byproduct of bitumen extraction (Wu et al., 2019). Nonpolar organics including NAs are considered to be responsible for the majority of the toxic effects of OSPW and for this reason they are among the most studied PRCs in OSPW. However, the majority of these toxicology studies have used technical NA mixtures that are available commercially. NAs are classically defined as $C_nH_{2n+Z}O_2$, where Z indicates hydrogen atoms lost to ring formation, and commercially available NA mixtures are comprised of classical NAs. The naphthenic acid fraction components (NAFCs) of OSPW differ from commercial mixtures in that they are more complex than NAs with higher proportions of ring structures and oxygen species (Bauer et al., 2019; Brunswick et al., 2017; Marentette et al., 2015; Vander Meulen, Schock, et al., 2021). Consequently, commercial NAs and NAFCs may have different toxicologic effects in some tissues (Bartlett et al., 2017; Loughery et al., 2019).

NAFCs are present in surface and ground water samples in the AOSR because bitumen is naturally present in rock formations and riverbeds, however, these concentrations are generally below 1mg/L. Wetlands near AOSR mining have reported concentrations between 8.1 and 36.1mg/L, whereas OSPW contains concentrations of NAFCs up to 110mg/L (Headley & McMartin, 2004; Vander Meulen, Klemish, et al., 2021; Vander Meulen, Schock, et al., 2021). However, these concentrations vary greatly depending on the measuring technique and there is not a standard and robust method of measuring NAFC concentrations (Brown & Ulrich, 2015). While the exact exposures of wildlife to NAFCs cannot be measured at this point, researchers have in some cases been able to use novel chemical markers to confirm that OSPW from tailings ponds have migrated into groundwater reaching the Athabasca River system (Hewitt et al., 2020). This evidence of seepage of OSPW into natural waterways in the AOSR suggest that aquatic animals are likely being exposed to greater quantities of NAFCs than appear in waterways from natural bitumen due to oil sands activity.

NAFCs and OSPW have been reported to have adverse effects on the reproductive health of animals. In one study treatment of Sprague Dawley rats with commercial naphthenic acids at doses of 300 and 900mg/kg/d significantly

reduced the number of live born offspring (McKee et al., 2014). Similarly, female Wistar rats that were given naphthenic acids isolated from AOSR OSPW (60 mg/kg/d, 5 days per week orally) had some reduced fertility but no malformations among offspring (Rogers et al., 2002). Other effects reported by Rogers et al. in these female rats were increased liver weights, elevated blood amylase and hypocholesterolemia and excessive hepatic glycogen accumulation (Rogers et al., 2002). In contrast to mammalian studies, a number of studies have investigated the effects of NA and OSPW on reproductive outcomes in fish species; reported adverse effects include reduced fertilization success, premature hatching, embryo deformities, and reduced expression of secondary sexual characteristics (He et al., 2012; Kavanagh et al., 2013; Li et al., 2017; Peters et al., 2007). In addition, larvae exposed to NAFC had skeletal deformities (Loughery et al., 2019) and OSPW has been shown to affect adipogenesis (Peng et al., 2016). Both osteoblasts (bone cells) and adipocytes (fat cells) share a common mesenchymal cell lineage. As bone strength can be affected by both the development of bone and the interface between bone and bone marrow adipose tissue (BMAT), any exposure that can affect the differentiation of these mesenchymal precursor cells into osteoblasts or adipocytes has the potential to influence bone strength. However, the effects of NAFCs from OSPW on osteoblastogenesis or adipogenesis are unknown.

1.1.2. Mesenchymal stem cells

Mesenchymal stem cells (MSCs) are pluripotent fibroblast-type stem cells that can differentiate into osteoblasts, adipocytes, chondrocytes, myocytes, and neurons (Pittenger et al., 1999). MSCs were first found in bone marrow in 1976 (Friedenstein et al., 1976) and since have been found in human umbilical cord (Wang et al., 2004), placenta (Antoniadou & David, 2016), adipose tissue (Casteilla et al., 2005) and in dental pulp (Leong et al., 2012). It is now well known that chemicals can influence MSC lineage commitment in vivo (Bateman et al., 2017). For example, treatment with rosiglitazone, a PPARy agonist, induces the differentiation of MSCs in bone into adipocytes and suppresses osteoblastogenesis resulting in an increase in bone marrow adiposity and reduced bone strength and quality (Ali et al., 2005). Bone marrow adiposity is associated with reduced bone quality and has been observed in most bone loss conditions (Bredella et al., 2011; Moerman et al., 2004). Differentiation path selection is determined by the biological, chemical and physical milieux surrounding the MSCs. These factors converge to affect transcription factors driving the differentiation of MSCs toward adipogenesis (C/EBPs and PPAR γ) or osteoblastogenesis (RUNX2 and TAZ) (Q. Chen et al., 2016).

1.1.3. Bone

There are five types of bone: long, short, flat, irregular, and sesamoid. The baculum is a long bone; long bones are typically longer than they are wide and have a rounded head (epiphysis) at both ends and a long shaft (diaphysis) that carries the load between the epiphyses. The diaphysis is typically loaded as a simply supported beam; its mechanical strength is a product of the properties of the material and its cross-sectional area.

Bone is a composite material made of mineral crystals bound to proteins. This construction combines the strength and resilience of the proteins with the stiffness and hardness characteristics of mineral. The mineral phase is comprised of crystals containing calcium and phosphate, called hydroxyapatite, and the protein phase consists of collagen (Glimcher, 2006). The collagen is assembled into fibers, and along with other proteins these are bound to the mineral crystals (Glimcher, 2006). To provide an optimal stiffness to mass ratio, bones are hollow with the dense outer cortical shell farthest from the neutral axis. The hollow interior contains a fine network of trabecular bone while the outer cortical shell provides the mechanical strength and attachment sites for tendons (Clarke, 2008). Bone strength is a product of BMD and matrix integrity and is maintained through the continuous process of remodeling. Bone remodeling involves the resorption of old bone matrix by osteoclasts and the deposition of new bone matrix by osteoblasts. This process both repairs microdamage to the bone and allows the bone to adapt to novel loading stimuli. The relative rate of deposition to resorption determines the rate of change of BMD (General (US), 2004).

Bone tissue is made up of three major cell types: osteoblasts, osteocytes, and osteoclasts. Osteoblasts are located on the osteoid and synthesize new bone matrix on bone-forming surfaces as they differentiate from osteoblast progenitor cells during the formation phase of the remodelling cycle. Osteoprogenitor cells are derived from multipotent mesenchymal stem cells (MSCs), which can also differentiate into adipocytes, chondrocytes, myocytes, and neurons. As the cells age, 60-80% of mature osteoblasts undergo apoptosis while the rest become embedded in the matrix and differentiate into osteocytes (Komori, 2016). Osteocytes are terminally differentiated osteoblasts and are embedded in the bone matrix. Osteocytes are the most plentiful cell type in bone and have mechanosensory and regulatory function; they can emit stress signals in response to changes in bone strain imposed by mechanical loading and react to endocrine signaling to regulate bone remodelling (K. Lee et al., 2003). Osteoclasts are multi-nucleated cells that perform the resorption phase of bone remodeling and are derived from hematopoietic mononuclear precursor cells of the monocytemacrophage lineage (Clarke, 2008). Osteoclast activation is dependent on osteocyte and osteoblast release of receptor activator of NF-kB ligand (RANKL) and macrophage colony stimulating factor (M-CSF)(Rauch et al., 2010). Once active, osteoclasts dissolve the mineral content of the bone by secreting H+ ions and secrete the cathepsin K enzyme that digests the protein content of the matrix (Komori, 2018). Bone remodeling is regulated by communication between osteocyte, osteoblast and osteoclast cells and controlled by local and systemic factors to fulfill both the structural and metabolic functions of bone (Hauge et al., 2001).

Initiation of remodeling is caused by signaling from osteocytes and begins with the activation phase. In targeted remodeling, osteocyte apoptosis leads to release of paracrine factors that increase local angiogenesis and recruitment of osteoclast and osteoblast precursors (Atkins & Findlay, 2012; H. Chen et al., 2015; Dallas et al., 2013; Tatsumi et al., 2007). Non-targeted remodeling occurs in response to hormones, allowing access to bone calcium stores, not directed to a specific site (Kenkre & Bassett, 2018). Next, during resorption osteoclasts pump protons generated by carbonic anhydrase II onto the bone surface being resorbed, dissolving the bone mineral (Kenkre & Bassett, 2018). The remaining collagen is degraded by proteases including cathepsin K and matrix metalloproteinases. Resorption ends by osteoclast programmed cell death (Kenkre & Bassett, 2018). The reversal phase follows resorption in which freshly resorbed bone surface is prepared for deposition in new bone matrix by osteoblastic cells that remove unmineralized collagen matrix. Signaling between osteoblasts and osteoclasts occurs to ensure no net bone loss (Howard et al., 1981; Sims & Martin, 2015). Osteoclasts express ephrinB2 and osteoblasts express EphB4 ephrin receptor, suggesting that the reversal phase may be controlled by bidirectional ephrin signaling from osteoblasts to osteoclasts (Zhao et al., 2006). Other signaling pathways including BMP-2, TGF-B and IGF-1 may also be involved (Sims & Martin, 2015; Zhao et al., 2006). Bone formation occurs in two steps: first osteoblasts synthesize and secrete a type 1 collagen-rich osteoid matrix, then osteoblasts play a key role in regulating osteoid mineralization (Atkins & Findlay, 2012). After mineralization is complete, osteoblasts undergo apoptosis, become bone-lining cells or terminally differentiate into osteocytes embedded in the bone matrix.

Few studies have investigated the effect of NAs or OSPW exposure on bone, and no studies have investigated the effect of NAs on bone metabolism, osteoclastic or osteoblastic activities. One study that reported on the effects of naphthenic acid fraction components (NAFC) from aged or fresh OSPW and commercial NAs on fathead minnow larvae reported bone deformities and altered transcriptomic networks underlying skeletal development. Skeletal gene networks of chondrocyte development and endochondral ossification were altered by NAFC and commercial NA. Spinal deformities were most pronounced in larvae exposed to aged NAFC, and transcriptomic profiles were more similar between fresh and aged NAFC than commercial NA (Loughery et al., 2019). In another study, exposure to the water-soluble fraction of light cycle oil that contains several PRCs resulted in lower bone mineralization in sea bass (Danion et al., 2011). Several groups have reported on the effects of NAs or OSPW on hormones that can affect bone strength. For example, exposure to 10 mg/L naphthenic acids extracted from discharge water from Statoil platform Oseberg C did not change cortisol production, but high doses (10µg dry weight m⁻¹L⁻¹) did increase 17B-estradiol and progesterone production and decrease testosterone production in human adrenocortical carcinoma (H295R)-cells, but lower doses (0.1 and 1µg dry weight $m^{-1}L^{-1}$) did not (Knag et al., 2013). Both estrogen and testosterone are important

for maintenance of bone health, and a decrease in testosterone in particular may result in lower bone strength (Khosla et al., 2012; Mohamad et al., 2016). Although Lister et al. (2008) found that goldfish exposed to OSPW but not naphthenic acid extract had increased plasma cortisol levels, which they suggest indicates an effect on glucocorticoid hormone biosynthesis, the serum cortisol data for the naphthenic acid extract treated goldfish was not presented in the paper (Lister et al., 2008). These studies suggest several potential mechanisms by which OSPW or NAFCs may inhibit bone strength, although little is known of their direct effect on MSCs or osteoblasts.

1.1.4. Bone marrow adipose tissue

Bone marrow adipose tissue (BMAT) comprises about 70% of bone marrow (BM) volume (Cawthorn et al., 2014) and is functionally distinct from the other two primary types of adipose tissue, white adipose tissue (WAT) and brown adipose tissue (BAT), in that it resists insulin- and cold-stimulated glucose uptake (Suchacki et al., 2020). There are two types of adipocyte populations that comprise BMAT: regulated BM adipocytes that are interspersed with haemopoietic BM and constitutive bone marrow adipocytes that cluster together at distal skeletal sites (Bukowska et al., 2018). The majority of BM adipocytes arise from leptin-receptor positive MSCs, a quiescent progenitor cell population that can differentiate primarily into either mature adipocytes or osteoblasts (Zhou et al., 2014). Marrow adiposity is correlated with low bone mass and the fate decision of MSC progenitor cells into adipocytes or osteoblasts appears to be dependent on peroxisome proliferator-activated receptor gamma (PPAR γ) activity; PPAR γ insufficiency increases bone formation and PPAR γ overexpression inhibits it (Akune et al., 2004; S. W. Cho et al., 2011). The primary regulatory transcription factors of adipogenesis and osteoblastogenesis, PPAR γ and runt-related transcription factor 2 (RUNX2), are reciprocally regulated BM MSCs (Stechschulte & Lecka-Czernik, 2017).

As with bone, few studies have reported on the effects of NAFCs or OSPW on adipocytes. One notable exception is a study by Peng et al. (2016) in which they identified components of OSPW as PPAR γ ligands and were able to induce PPAR γ -mediated adipogenesis in murine 3T3-L1 adipocyte progenitor cells treated with OSPW (Peng et al., 2016). Peng et al. also investigated the effects of fractionated OSPW and found that while the nonpolar fraction, which contained most NAs, did contain PPARy ligands, it did not induce adipogenesis to a similar degree as the other fractions of OSPW (Peng et al., 2016). The activation of PPARy by other environmental contaminants has been shown to lead to the induction of adiposity in animals (Fang et al., 2015; Feige et al., 2007; Harada et al., 2015; Pillai et al., 2014). In fact, a study that investigated the effect of exposure to the wastewater from fracking, another oil extraction method, found an induction of PPAR γ -mediated adipogenesis (Kassotis et al., 2018). NAFCs also inhibit mitochondrial respiration, uncouple oxidative phosphorylation and increase reactive oxygen species emission in isolated mitochondria harvested

from trout liver (Rundle et al., 2018). This impaired mitochondrial function could affect oxidative metabolism in adipocytes, impairing their ability to generate energy to support important metabolic pathways like triglyceride synthesis and gluconeogenesis. Taken together these studies suggest that NAFCs may have an impact on adipocyte function as well as differentiation.

1.1.5. Pathways regulating MSC differentiation

The differentiation of MSCs into mature osteoblasts or adipocytes is regulated by many signaling pathways at different stages of development, but the focus of this section are two pathways that have substantial impacts on MSC differentiation and are likely targets for modulation by exogenous chemicals. Exogenous glucocorticoids are well known to induce a loss in BMD (Briot & Roux, 2015) and many endocrine disrupting chemicals affect either serum cortisol levels or act via the glucocorticoid or mineralocorticoid receptors (Diamanti-Kandarakis et al., 2009; Zhang et al., 2019). Secondly, as discussed above, activation of PPARy is capable of inducing differentiation of osteoblast progenitor cells to an adjocyte phenotype (E.-S. Cho et al., 2012). Importantly OSPW has previously been found to contain PPARy ligands (Peng et al., 2016; Stechschulte & Lecka-Czernik, 2017). Therefore, it is biologically plausible that if NAFC can activate either the glucocorticoid receptor (GR) or PPARy pathways in MSCs there could be an inhibition of osteoblastogenesis or an increase in bone marrow adipose tissue; both of which have the potential to impair overall bone strength.

Glucocorticoids have adverse effects on skeletal health, and glucocorticoid-induced osteoporosis is the most common form of secondary osteoporosis(Briot & Roux, 2015). Secondary osteoporosis is the loss of bone mineral density from any clinical condition other than aging (Briot & Roux, 2015). Glucocorticoids exert their effect on the skeletal system through both direct effects on bone and indirectly by modulating the neuroendocrine system, calcium metabolism and inducing muscle myopathy (Canalis et al., 2007). Glucocorticoids have direct effects on each of the three major bone cell types, inhibiting differentiation and function and inducing apoptosis in osteoblasts, inhibiting the function and inducing apoptosis in osteocytes, and by increasing osteoclastogenesis and reducing osteoclast apoptosis (Canalis et al., 2007). Although low levels of GR activity induced by endogenous glucocorticoids is required for bone formation (Sher et al., 2004), exogenous glucocorticoids reduce both bone mass and quality through reduced bone formation and increased resorption. Glucocorticoids inhibit osteoblastogenesis by regulating the nuclear factors of the CCAAT/enhancer binding protein (C/EBP) family and the peroxisome proliferator activated receptor gamma 2 (PPAR γ 2) and inhibiting the transcriptional repression of preadipocyte factor-1 (Pref-1)(Canalis, 2005). Glucocorticoids also inhibit genes involved in bone formation such as collagen type 1 (Col1a1) and Runx2, or through antagonizing bone morphogenic protein-(BMP-) and wingless/integrated- (Wnt-) signaling pathways (Canalis et al., 2007). Glucocorticoids also inhibit osteoblast function by suppressing the expression of

15

insulin-like growth factor 1 (IGF-1) which is required for collagen synthesis (Canalis, 2005). These actions are mediated through the activation of GR, a nuclear receptor that alters gene expression by binding as dimers to GCresponsive elements in genes like nuclear factor kappa-light-chain-enhancer of activated B cells (NF-kB), activator protein 1(AP-1), interferon regulatory transcription factor 3 (IRF-3) or signal transducer and activator of transcription 5 (STAT5) (Kassel & Herrlich, 2007), or through monomeric GR activity (Rauch et al., 2010). Glucocorticoids also increase osteoclastic bone resorption by increasing RANKL expression and reducing OPG expression by osteoblasts. Enhanced bone resorption is only transient; prolonged exposure lowers osteoclast numbers and resorption (Canalis & Delany, 2002; Henneicke et al., 2014; Mitra, 2011). These data suggest that glucocorticoids induce BMD decrease primarily by acting on osteoblasts. The glucocorticoid dexamethasone down-regulates osteocalcin in bone cells (S.-M. Chen et al., 2018) and in osteogenic media strongly induces adipogenesis in bone marrow stromal cells, in part via induction of PPARy (Ghali et al., 2015). In fact GR and PPARy cross-talk is largely synergistic, particularly in the case of lipogenesis where GR is acting via PPAR γ (Hasan et al., 2018).

PPARγ is the primary transcription factor regulating adipogenesis and *in vivo* exposures to PPARγ agonists have been shown to induce adipogenesis and suppress osteogenesis in mouse BM-derived MSCs (Watt & Schlezinger, 2015). Increased bone marrow adiposity is associated with lower bone strength and

quality (Bredella et al., 2011; Moerman et al., 2004). PPARy is part of the nuclear receptor superfamily that includes steroid, retinoid an thyroid hormone receptors, and is most highly expressed in adipose tissue (Mangelsdorf et al., 1995). In addition to regulating adipocyte differentiation, PPARy plays key roles in lipid storage and mobilization, glucose metabolism and inflammatory responses (Janani & Ranjitha Kumari, 2015). PPAR γ also plays a crucial role in regulating MSC differentiation. PPARy not only induces the expression of adipogenic genes, but osteoblast differentiation also requires the inhibition of PPARy as PPARy inhibits the expression of genes required for osteogenic differentiation (Wan, 2010). There is evidence that exposure of MSCs to chemicals that act as PPARy agonists can promote adipogenesis and reduce bone formation (Watt & Schlezinger, 2015). Importantly studies have shown that compounds in OSPW or other oil-extraction wastewater can induce PPARy activity and subsequently increase adipogenesis (Kassotis et al., 2018; Peng et al., 2016). However, it remains to be elucidated whether the induction of PPARy activity by compounds in OSPW can have an effect on the bone strength through the differentiation of progenitor cells into osteoblasts.

1.2. Main Objective

The differentiation of progenitor cells into osteoblasts is central to the ability of bone to perform its primary structural function, adapt to loads, and 17

repair damage throughout the life-span of mammals. Osteoblasts originate from the same pool of progenitor cells as bone marrow adipose tissue, and the differentiation into mature osteoblasts is influenced by many endocrine factors including glucocorticoids and sex hormones. Consequently, NAFCs of OSPW, which contain endocrine disrupting chemicals, may be pushing progenitor cells away from osteoblastic differentiation and toward adipogenesis. This may then contribute to the lower bone strength observed in free-living river otters trapped in the AOSR. Therefore, the overall goal of this thesis was to test the hypothesis that exposure to NAFC can alter the differentiation of MSC progenitor cells to favour reduced bone formation and bone quality. Specifically, this thesis examined the effects of NAFCs on osteoblast differentiation and adipocyte differentiation and sought to identify mechanisms by which these effects occur to further our understanding of the impact OSPW related contaminants are having on wildlife in areas affected by mining activity.

2. Chapter 2

Prepared for publication.

2.1. Preface and significance to thesis

Our lab and collaborator Dr. Phillipe Thomas found that there was a decrease in baculum strength associated with proximity to oil sands mining activity in north American river otters (Lontra canadensis) (Thomas et al., 2021). However, the contribution of NAFCs to this effect and the mechanism by which PRCs were impairing bone strength was unknown. We had also received the naphthenic acid fraction of OSPW collected from a tailings pond from Dr. Richard Frank and Dr. Mark Hewitt from Environment and Climate Change Canada. Thomas et al. 2021 were not able to measure NAFC exposure in their study, but as they are considered to have significant toxicity, we examined the effects of NAFCs on osteoblast development and function. This work demonstrates that NAFC can inhibit osteoblast differentiation, which is required for bone formation, and acts at least in part via GR activity. Interestingly the inhibition of osteoblast differentiation was greater with NAFC treatment than with the GR activator positive control dexamethasone. This work demonstrates that exposure to NAFCs may impair bone strength and this effect should be investigated in *in vivo* models.

2.2. Author contribution

RMG, PJT and ACH were responsible for the experimental design of this study. LMH and RAF provided the NAFC stock solution. RMG was the primary contributor to the data in Figure 1, Figure 2, Figure 3, and Figure 4. LJ helped with preliminary lab experiments. RMG wrote the first draft of the manuscript and the rest of the authors provided edits and feedback on the manuscript.

2.3. Naphthenic acid fraction components from oil sands process affected water from the Athabasca Oil Sands Region impair murine osteoblast differentiation and function

Naphthenic acid fraction components from oil sands process affected water from the Athabasca Oil Sands Region impair murine osteoblast differentiation and function

Robert M. Gutgesell¹, Laiba Jamshed¹, Richard A. Frank², L. Mark Hewitt², Philippe J. Thomas³ Alison C. Holloway¹

- 1. Department of Obstetrics and Gynecology, McMaster University, Hamilton, ON, Canada
- 2. Water Science and Technology Directorate, Environment and Climate Change Canada, Burlington, ON, Canada
- 3. Environment and Climate Change Canada, National Wildlife Research Centre, Ottawa ON., Canada

Address for correspondence:	Dr. Alison Holloway
-	Department of Obstetrics & Gynecology
	McMaster University
	RM HSC-3N52
	1280 Main Street West
	Hamilton, Ontario, Canada, L8S 4K1
	Phone: (905) 525-9140 ext. 22130
	E-mail: hollow@mcmaster.ca

Short title: Naphthenic Acid Fraction Components Inhibit Osteoblastogenesis

Key words: Naphthenic acids, oil sands process-affected water, osteoblast, osteoblastogenesis, osteoblast differentiation, glucocorticoid receptor

Abstract

The extraction of bitumen from surface mining in the Athabasca Oil Sands Region (AOSR) produces large quantities of Oil Sands Process-affected Water (OSPW) that need to be stored in settling basins near the extraction sites. Compounds in OSPW impair bone health in some organisms, which can lead to increased fracture risk and lower reproductive fitness in wildlife. Naphthenic acid fraction components (NAFCs) are among the most toxic class of compounds in OSPW, however the effect of NAFCs on osteoblast development is largely unknown. In this study we demonstrate that NAFCs from OSPW inhibit osteoblast differentiation and deposition of extracellular matrix which is required for bone formation. Extracellular matrix deposition was inhibited in osteoblasts exposed to 12.5-125 mg/L of NAFC for 21 days, concentrations that had no effect on cell viability. We also show that components within NAFCs inhibit the expression of gene markers of osteoblast differentiation and function, namely alkaline phosphatase, osteocalcin and collagen type 1 alpha 1 (*Colla1*). These effects were partially mediated by the induction of glucocorticoid receptor activity; NAFCs induce the expression of the GR activity marker genes Sgk1 and p85a, and inhibits GR protein and Opg RNA expression. This study provides evidence that NAFC concentrations \geq 12.5mg/L can directly act on osteoblasts to inhibit bone formation and suggests that NAFCs contain components that can act as GR agonists which may have further endocrine disrupting effects on exposed wildlife.

Introduction

The Athabasca Oil Sands Region (AOSR) is one of the world's largest deposits of extractable bitumen (Alberta Energy Regulator, 2015). The process of bitumen extraction uses large amounts of water that becomes OSPW; this OSPW is then stored in tailings ponds and cannot be discharged into receiving waters after use (Wu et al., 2019). There is concern that chemicals from bitumen extraction may be affecting development and reproductive health of animals in areas impacted by mining activity; in particular many of these concerns are centred around the toxicity of oil sands process affected water (OSPW). However, there is concern that components of OSPW can migrate from tailings ponds into nearby wetlands through groundwater mixing or seepage from tailings ponds.

OSPW contains a variety of compounds including naphthenic acid fraction components (NAFCs) which are a large and diverse group of organic acids that are natural components of bitumen (Clemente & Fedorak, 2005). NAFCs are considered the most toxic class of organic compound in OSPW (C. Li et al., 2017). As a result, the toxicity of NAFCs has been widely studied. Indeed, there is a considerable amount of evidence demonstrating the toxicity of OSPW on
invertebrates (Bartlett et al., 2017) and fish (Mahaffey & Dubé, 2016; Marentette et al., 2015, 2017). Although few studies have investigated the effects that chronic exposure to sub-toxic concentrations of NAFCs have on major organ systems in mammals (Clemente & Fedorak, 2005; C. Li et al., 2017), there is considerable evidence that exposure to OSPW and NAFC can act as endocrine disrupting compounds. Sub-toxic effects of NAFCs that have been observed include alterations in sex hormone and glucocorticoid production (Knag et al., 2013; Lister et al., 2008), impaired development and reproductive success and reduced secondary sexual characteristics in animals exposed to NAFC (Kavanagh et al., 2013; C. Li et al., 2019; Philibert et al., 2019; Rogers et al., 2002).

More recently, in a study investigating the baculum strength of animals in the AOSR, lower baculum strength was observed in river otters (*Lontra canadensis*) collected in areas with high levels of mining activity compared to those collected in areas far from mining activity (Thomas et al., 2021). A decrease in baculum strength can reduce the reproductive fitness of male mammals, thereby presenting a risk to the stability of these wild populations (Brassey et al., 2018; Sonne et al., 2015; Stockley et al., 2013). Previous studies have also shown that exposure to endocrine disrupting chemicals lower baculum bone mineral density (BMD) (Sonne et al., 2015). Although, there is limited information regarding the effects of OSPW or NAFC on mammalian bone, a study in fathead minnow larvae reported that both a commercial technical naphthenic acid mixture and NAFC of tailings water from the AOSR induced skeletal deformities and altered gene networks related to chondrocyte development and endochondral ossification (Loughery et al., 2019). These data suggest that it is biologically plausible that NAFC may also impact mammalian bone development.

Bone strength is maintained throughout life by the process of remodeling, which allows bone to adapt and repair microdamage through the resorption of old bone and deposition of new bone matrix (Seeman, 2003). This process is regulated by endocrine factors and paracrine signaling between osteocytes, osteoblasts and osteoclasts (Hauge et al., 2001). Osteoblasts synthesize new bone matrix on bone-forming surfaces as they differentiate from osteoblast progenitor cells during the formation phase of the remodelling cycle (Komori, 2016). The primary endocrine pathways regulating BMD are parathyroid hormone (PTH) (Silva & Bilezikian, 2015; Stein et al., 2013), insulin-like growth factor 1 (IGF-1)(McClung et al., 2014; Plotkin & Bellido, 2016) and glucocorticoids (Rauch et al., 2010; Weinstein et al., 1998). Glucocorticoid receptor (GR) activation in osteoblasts inhibits differentiation and production of extracellular matrix (ECM) (Rauch et al., 2010). Further, the use of synthetic glucocorticoids like dexamethasone is also the most common form of secondary osteoporosis (Briot & Roux, 2015). Exposure to OSPW has been found to increase serum cortisol levels (Lister et al., 2008), but no studies have investigated whether chemicals in OSPW act as agonists for PTH, IGF1 or glucocorticoid receptors. The goals of this study were to 1) investigate the effects of the NAFC extracted from OSPW from an active mining operation within the AOSR (Frank et al., 2006) on the osteogenic

potential of mouse pre-osteoblastic (MC3T3-E1) cells and 2) determine if the underlying mechanism is glucocorticoid receptor dependent.

Materials & methods

NAFC preparation

Naphthenic acid extracts were isolated from tailings pond water collected from Syncrude Canada Ltd. West In-pit settling basin in Fort McMurray, Alberta, Canada in 2009, using a procedure described in (Frank et al., 2006). An NAFC stock solution was prepared and stored in glass vessels in darkness and preserved in 0.05 M NaOH with a final concentration of 2504 mg/L determined via liquid chromatography/quadrupole mass spectrometry with time-of-flight detection (LC/QTofF) as previously described in Marentette et al., (2017).

MC 3T3-E1 cell culture & treatment

MC 3T3-E1 subclone 4 mouse newborn fibroblasts (ATCC, Manassas, VA, USA) were maintained in ascorbic acid-free alpha minimum essential medium (aMEM) with ribonucleosides, deoxyribonucleosides, 2mM L-glutamine and 1 mM sodium pyruvate (Gibco, Waltham, MA, USA), supplemented with 10% fetal bovine serum (Hyclone, Logan, UT, USA) and 1% P/S. At 70% confluence cells were passaged to 6-well plates and left to reach confluence (100%). 48hrs after reaching 100% confluence (day 0), differentiation was induced with 100µM L-ascorbic acid (EMD, Mississauga, ON, Canada) and 2mM

 β -glycerophosphate (Sigma, St. Louis, MO, USA) (Wang et al., 1999) added to the supplemented aMEM media (differentiation media). To investigate whether NAFC inhibits osteoblastogenesis, cells were treated beginning on day 0 with 125 mg/L, 25 mg/L, 12.5 mg/L and 1.25 mg/L of the NAFC mixture that was diluted in differentiation media (described above). These concentrations encompass the range of NAFCs that have been reported in surface waters, wetlands, and tailings ponds in the AOSR (Vander Meulen et al., 2021). Dexamethasone (1µM, Sigma), a glucocorticoid receptor agonist, was used as a positive control to assess the effects of GR activation during differentiation (Rauch et al., 2010). The pH of the media was adjusted to be similar across all groups using NaOH or HCl after the addition of NAFC. Cells were treated for 7, 14, or 21 days after the initiation of differentiation; media was changed every 4 days.

Cell viability

Cells were seeded and treated in 96-well plates following the protocols described above. After 7 and 14 days of treatment, 10% alamarBlue reagent (BioRad Laboratories, Hercules, CA, USA) was added to fresh media and cells were incubated for 2h. Absorbance was then read at 570nm in a plate reader (Synergy H1 microplate reader, Norgen BioTek, Thorold, ON, Canada) with a reference read at 600nm. Treatments were considered cytotoxic if the average absorbance for the group was below 80% of the control group.

ALP staining and enzyme activity assay

Alkaline phosphatase (ALP) is an early marker of osteoblast differentiation (Lee et al., 2017). To visualize ALP abundance, after 14 days of treatment cells were washed with PBS and fixed in 10% formalin for 10 min. The fixed cells were incubated in a solution containing 1mg/ml Fast Blue RR salt and 0.4 mg/mL naphthol AS-MX phosphate disodium salt (Thermo Fisher Scientific, Waltham, MA, USA) at pH 8.4 for 15 min. The reactions were stopped by rinsing with deionized water and cells were photographed under a light microscope (Motic Microscopes, Xiamen, China). To measure ALP activity, an enzymatic ALP activity assay (Abcam, Cambridge, UK) was used according to the instructions provided by the manufacturer. Briefly, cells were harvested with a cell lysis solution, a p-nitrophenyl phosphate (pNPP) phosphatase substrate was applied to the samples, and absorbance was read at 405nm using a microplate reader (Norgen BioTek). Absorbance data was corrected to background and compared to control.

Alizarin Red S staining of MC3T3-E1 osteoblasts

To measure calcium deposition, after 21 days of treatment, cells were washed with Dulbecco's phosphate-buffered saline (DPBS), then fixed with 10% formalin and stained with 2% Alizarin Red S (ARS) solution (EMD) for 15 min. The ARS solution was removed, and the cells were then washed 5x with water. Representative images were taken using phase-contrast microscopy (Motic Microscopes). The dye was extracted using 10% acetic acid, cells were separated, and supernatant was transferred to a 96 well plate and absorbance was read at 405nm in a microplate reader (Norgen BioTek). Absorbance data was corrected for background and compared to control.

Real-time quantitative PCR

After 7, 14 or 21 days of treatment, cells were washed with phosphatebuffered saline (PBS) and RNA was extracted using TRIzol reagent (Invitrogen, Waltham, MA, USA). RNA concentrations were determined using the NanoDrop One Microvolume UV-Vis Spectrophotometer (Thermo Fisher Scientific). Complementary DNA (cDNA) was made from 2 µg of mRNA using the High-Capacity cDNA Reverse Transcription Kit (Applied Biosystems, Waltham MA, USA) as per the manufacturer's instructions. Real-time quantitative PCR (RTqPCR) was carried out using PerfeCTa SYBR Green FastMix (Quanta Biosciences, Gaithersburg, MD, USA) on the CFX384 Touch Real Time PCR Detection System (Bio-Rad Laboratories). The PCR cycling settings included polymerase activation (95 °C for 10 m), followed by 40 cycles of denaturing (95 °C for 15s) and annealing/elongation (60 °C for 1m). Levels of gene expression were generated using the $2(-\Delta\Delta C(T))$ method (Livak & Schmittgen, 2001) and normalized using the geometric means of two reference genes: Tyrosine 3monooxygenase/tryptophan 5-monooxygnase activation protein zeta (Ywhaz) and peptidylprolyl isomerase A (*Ppia*). Gene targets for osteoblast differentiation:

alkaline phosphatase (*Alp*), collagen type 1 alpha 1 chain (*Col1a1*), *Osteocalcin*, and runt-related transcription factor 2 (*Runx2*), glucocorticoid receptor activity: serum/glucocorticoid regulated kinase 1 (*Sgk1*), receptor activator of nuclear factor kappa-B ligand (*Rank1*), osteoprotegerin (*Opg*) and phosphatidylinositol 3kinase regulatory subunit alpha (*p85a*) were measured. Primer sequences are in Table 1.

Immunoblotting

After 21 days of treatment, cells were isolated in lysis buffer (50 mM HEPES, 150 mM NaCl, 100 mM NaF, 10 mM sodium pyrophosphate, 5 mM EDTA, 250 mM sucrose, 1 mM dithiothreitol and 1 mM sodium orthovanadate, with 1% Triton X and one tablet of cOmplete Protease Inhibitor Cocktail (Roche, Basil, Switzerland) per 50 ml) then stored at -80 °C until analysis. Protein concentration was determined with the Pierce BCA Protein Assay kit (Thermo Fisher Scientific). Lysates were then diluted with Laemmli buffer and run on a 10% polyacrylamide gel (Bio-Rad Laboratories) to separate proteins based on size. Samples were then transferred to a polyvinylidene difluoride (PVDF) membrane and blocked in 5% BSA for 1h at room temperature (20–25 °C). Membranes were incubated with primary antibody (1:1000) overnight at 4 °C. Appropriate secondary antibodies were used at a concentration of 1:10000. Bound antibodies were detected using Clarity Western ECL Substrate (BioRad Laboratories). For primary antibody information see Table 2.

Statistical analysis

Data are expressed as the means +/- standard errors of the means (SEM). Statistical analyses were carried out using GraphPad Prism 9.2 (GraphPad Software Inc, San Diego, CA, USA). One-way analysis of variance (ANOVA) was followed by post-hoc testing (Dunnett multiple comparisons) when significance was indicated (p<0.05).

Results

NAFC inhibits osteoblastogenesis in murine pre-osteoblasts

Treatment with all doses of NAFC for 8 and 14 days had no impact on the viability of MC3T3-E1 pre-osteoblast cells (Fig. 1). Following 14 days of treatment with NAFC, ALP activity was assessed by ALP activity assay and staining. Treatment with 125mg/L NAFC as well as 1µM dexamethasone led to significantly lower ALP activity than in control cells (Fig. 2a-b). The deposition of calcium in a mineralized ECM is a common functional marker for the late stages of osteogenic differentiation. We measured the deposition of calcium in 3T3-E1 cells following 21 days of treatment with NAFC using Alizarin Red-S (ARS) calcium staining. Treatment with 12.5-125 mg/L of NAFC resulted in a dose-dependent reduction in extracellular calcium deposition (Fig. 2c-d).

To further evaluate the effect that NAFC has on osteoblastogenesis, the expression of markers of osteoblast differentiation *Runx2*, *Alp, osteocalcin* and *Col1a1* were measured after 7-, 14- and 21-days of exposure. NAFC had no impact on *Runx2* gene or protein expression at any time point (Fig. 3a, f), but the 12.5-125mg/L doses of NAFC induced significant reductions in the expression of *Alp, osteocalcin* and *Col1a1* genes at the 14- and 21-day time points relative to control treated cells (Fig. 3b-d). The lowest dose of NAFC significantly inhibited *Alp* gene expression at 7 days only. At 14 and 21 days, NAFC exposures resulted in significantly lower Alp expression at 12.5, 25 and 125mg/L doses (Fig. 3b). Similarly, 125mg/L of NAFC significantly inhibited *osteocalcin* expression at all

time points, and by 21 days all doses of NAFC had significantly lower *osteocalcin* expression (Fig. 3c). Accordingly, we observed significantly lower osteocalcin protein expression in cells exposed to 125mg/L of NAFC than control cells (Fig. 3e). *Col1a1* expression was unchanged after 7 days of NAFC treatment, but after 14 days cells treated with 12.5mg/L of NAFC and higher exhibited significantly lower Col1a1 expression than control cells, which then persisted to 21 days in cells treated with 125mg/L of NAFC (Fig. 3d).

NAFC inhibition of osteoblastogenesis is mediated by glucocorticoid receptor activity

Given that GR activity has a common and strong inhibitory effect on osteoblastogenesis, we sought to determine if GR was implicated in the mechanism by which NAFC inhibits osteoblastogenesis. We measured the gene expression of downstream markers of GR activity *Opg*, *Rankl* after 14 days of NAFC treatment and *Sgk1* and *p85a* after 14 and 21 days of treatment. We observed a dose-depended decrease in *Opg* expression as well as a significant increase in *Rankl* expression in the 125mg/L NAFC treated cells, resulting in a significant increase in the *Rankl/Opg* ratio at 14 days (Fig. 4a-c). We also observed significant increases in *Sgk1* gene expression after both 14 and 21 days of treatment with 125mg/L of NAFC, and after 14 days of treatment with 1.25mg/L of NAFC (Fig. 4d-e). *P85a* expression was significantly higher with 14 days of NAFC treatment in a dose-dependent manner, an effect which persisted at 21 days (Fig. 4f-g). We then measured protein markers of GR activity after 21 days of treatment with NAFCs and found that there was no change in OPG or RANKL protein expression with NAFC or the 1μ M dexamethasone GR agonist positive control, but there was a significant decrease in GR protein expression in cells treated with both 125mg/L of NAFC and dexamethasone (Fig. 4h-l). These results are consistent with an induction of GR activity by NAFC components functioning as GR agonists.

Discussion

Although there are numerous studies that have investigated the toxicological effects of environmental contaminants derived from oil and gas production (C. Li et al., 2017; Rodríguez-Estival & Smits, 2016), few of these have focused on bone health. Two studies have found skeletal deformities caused by OSPW or light oil fraction on bone in fish (Danion et al., 2011; Loughery et al., 2019) and another found that river otters in parts of the AOSR where mining activity is prevalent have weaker baculums than otters farther from mining activity (Thomas et al., 2021). In the present study we found that treatment with NAFCs derived from the AOSR are sufficient to inhibit osteoblast differentiation, determined by measuring calcified ECM, ALP activity and gene and protein markers of osteoblastogenesis. We also found NAFC exposure results in altered expression of components of GR signalling similar to what is seen with a known GR agonist dexamethasone.

In the current study, we found that treatment with NAFC inhibited the differentiation of precursor cells into mature osteoblasts. Treatment with 12.5-125mg/L of NAFC significantly decreased the deposition of extracellular matrix in differentiating osteoblasts (Fig. 2d), indicating an inhibition in the differentiation and primary function of these cells. These findings are in line with previous work in which exposure to the water-soluble fraction of light cycle oil, which contains several petroleum related compounds, resulted in lower bone mineralization in sea bass (Danion et al., 2011). Further, research from other

groups has demonstrated that commercial NA mixtures and NA fraction components of OSPW increase spinal deformities and skeletal gene networks of chondrocyte development and endochondral ossification, osteoclast activation and function in fathead minnow larvae (Loughery et al., 2019). Similarly, the data presented here demonstrates that NAFCs have a direct effect on osteoblasts.

In this study we also show that NAFC can activate GR in a manner similar to dexamethasone. Glucocorticoids like dexamethasone inhibit osteoblast mineral deposition and ALP activity through inducing GR receptor activity (Rauch et al., 2010). NAFC and dexamethasone treatment resulted in similar changes in the expression of GR protein, Opg and p85a RNA (Fig. 4a,f), suggesting NAFC is also acting as a GR agonist (Kuo et al., 2012; Xu et al., 2019). However, compounds in the NAFC mixture were less potently inducing Sgk1 expression to a lower degree than to dexame thas one (p=0.0119) (Fig.4e). This is despite the fact that the inhibition of ECM deposition was more profound in cells treated with NAFC vs dexamethasone (92.7% lower ECM deposition with 125mg/L NAFC treatment than CTL vs 25.3% lower ECM deposition with dexamethasone treatment than CTL, p<0.0001) (Fig. 2d). Previous studies that have investigated the effects of NAFCs on gene markers of skeletal development did not measure the expression of the markers of osteoblastogenesis Alp, Collal or osteocalcin (Loughery et al., 2019). The lower expression of Alp, Collal and osteocalcin that we observed (Fig. 3b-d) are consistent with what others have reported in osteoblasts or mesenchymal stem cells treated with glucocorticoids (Chen et al.,

2018; H. Li et al., 2015). Due to the fact that the NAFC is a mixture of several thousand organic acids, various components of the NAFC mixture could be functioning through multiple pathways that contribute to the totality of responses observed with the whole mixture. It is likely that some components of the NAFCs may be inhibiting osteoblast differentiation through other mechanisms in addition to the those acting via the GR pathway. NAFC from other sources have previously been shown to alter other pathways that can affect BMD, including altered sex hormone function (Knag et al., 2013; Lister et al., 2008) and increased inflammatory cytokine expression (Garcia-Garcia et al., 2011); the effects of our NAFC mixture on these pathways remains to be determined. Our data suggest that NAFC is acting through GR, but NAFC is a complex mixture and we have not ruled out whether it is also acting through PTH, wingless-related integration site (Wnt) or peroxisome proliferator-activated receptor gamma (PPAR_γ), or other pathways that can inhibit osteoblastogenesis.

Although our findings are consistent with the hypothesis that NAFC exposure is inhibiting osteoblast differentiation and function, our observations were in a cell line and thus further study examining the effect of NAFC exposure on bone *in vivo* are required. A major limitation of studies investigating the toxicological effects of NAFC is the challenge of measuring the tissue concentrations and compositions of NAFC in wildlife. However, the concentrations of NAFCs used in this study are similar to those used in other toxicological studies (Bartlett et al., 2017) and encompass the range of concentrations reported in surface water, industrially-affected wetlands and tailings ponds in the AOSR (Vander Meulen et al., 2021). Finally, while we found evidence of GR activity being induced by NAFC, we have not confirmed that this is the only mechanism of action.

In summary, we have shown that exposure to NAFCs has the potential to inhibit bone formation via inhibition of osteoblast differentiation. If our findings are confirmed *in vivo*, NAFC derived from OSPW may reduce bone strength of chronically exposed mammals which could compromise fitness and reproductive health.

Funding source

This study was supported by a CIHR Project Grant (PJT 159651) to ACH.

Declaration of Competing Interest

None.

Acknowledgements

We thank Annia A Martínez-Peña and Ahmed Ayyash (McMaster University) for their technical help with some of the cell culture experiments.

Disclaimer

The views in this paper are only held by the authors and are not representative of the official policy of the authors' individual affiliations.



Figures

Figure 1: Effects of NAFC on cell viability.

Cell viability of MC3T3-E1 osteoblastic cells after 8 and 14 days of treatment with of concentrations of NAFC from 1.25 mg/L to 125 mg/L and 1 μ M dexamethasone positive control.

Each value is the mean +/- SEM (n=6). *p < 0.05, compared with the corresponding control group at each time point.



Figure 2. NAFC inhibits osteoblast differentiation in MC3T3-E1 cells in a dose dependent manner.

(A) ALP staining representative images in MC 3T3-E1 cells treated for 14 days with 1.25-125 mg/L NAFC and 1µM dexamethasone positive control.

(B) ALP activity of MC 3T3-E1 cells treated for 14 days with 1.25-125 mg/L NAFC and 1µM dexamethasone positive control.

(C-D) ARS calcium staining assay results and representative images in MC 3T3-E1 cells treated for 21 days with 1.25-125mg/L NAFC and 1 μ M dexamethasone positive control.

Each value is the mean +/- SEM (n=6). *p < 0.05, **p < 0.01, ***p < 0.001, ****p < 0.001, #p < 0.05 positive control, the NAFC treated group compared with the differentiation control group.



Figure 3. Effects of NAFC on osteoblast markers during differentiation of MC3T3-E1 cells.

(A-D) Expression of osteoblastic marker genes *Runx2*, *Alp*, *osteocalcin* & *Col1a1* in MC 3T3-E1 cells treated for 7, 14 and 21 days with 1.25-125mg/L NAFC and 1 μ M dexamethasone positive control.

(E-G) Protein concentration and western blot representative images of osteocalcin and RUNX2 in MC 3T3-E1 cells treated for 21 days with 1.25-125mg/L NAFC and 1 μ M dexamethasone positive control.

Each value is the mean +/- SEM (n=6). p < 0.05, p < 0.01, p < 0.001, p < 0.001, p < 0.001, p < 0.05 positive control, the NAFC treated group compared with the differentiation control group.



Figure 4. NAFC induce glucocorticoid receptor activity in MC3T3-E1 cells.

Expression of glucocorticoid receptor acivity markers in MC3T3-E1 cells treated with 1.25-125mg/L NAFC for 14 and 21 days with 1 μ M dexamethasone positive control.

(A-C) RNA levels of *Opg* and *Rankl*, and *Rankl/Opg* ratio after 14 days of NAFC treatment.

(D-G) RNA levels of *Sgk1* and *p85a* after 14 and 21 days of NAFC treatment.

(H-J) Protein levels of OPG, RANKL and RANKL/OPG ratio following 21 days of NAFC treatment.

(I) Protein levels of GR protein following 21 days of NAFC treatment.

(J) Western blotting representative images.

(K-L) RNA levels of *Lgr5* after 14 and 21 days of NAFC treatment.

Each value is the mean +/- SEM (n=6). p < 0.05, p < 0.01, p < 0.01, p < 0.001,

****p < 0.0001, #p < 0.05 positive control, the NAFC treated group compared with the differentiation control group.

Table 1: Table of primers.

Gene	Forward/ Reverse	Sequence	Accession No.
ALP	Forward	5'-AGCAGGTTTCTCTCTTGGGC-3'	NM_007431.3
ALP	Reverse	5'-GGTGCTTTGGGAATCTGTGC-3'	
COL1A1	Forward	5'-GGTGGGGTGGGAAGGAATTT-3'	NM_007742.4
COL1A1	Reverse	5'-GGTCTAGGGAGCATCTCAGC-3'	
IGF1	Forward	5'-CAGGAGGGTGCAACATCAGA-3'	NM_010512.5
IGF1	Reverse	5'-GTGGCATCCCAGTGACAGAT-3'	
LGR5	Forward	5'-CAAGCCCTGTGAGCACCTAT-3'	NM_010195.2
LGR5	Reverse	5'-TCTGAACACGGTCAAAGCCA-3'	
OPG	Forward	5'-GGAACCCCAGAGCGAAACAC-3'	NM_008764.3
OPG	Reverse	5'-GCCAAATGTGCTGCAGTTCG-3'	
Osteocalcin	Forward	5'-TTCTGCTCACTCTGCTGACC-3'	NM_007541.3
Osteocalcin	Reverse	5'-TATTGCCCTCCTGCTTGGAC-3'	
p85a	Forward	5'-CAAAGCGGAGAACCTATTGC-3'	NM_001024955.2
p85a	Reverse	5'-ATAGCAGCCCTGCTTACTGC-3'	
PPIA	Forward	5'-GACAAAGTTCCAAAGACAGCAGAA-'3	NM_008907.2
PPIA	Reverse	5'-CCAAATCCTTTCTCTCCAGTGC-3'	
RANKL	Forward	5'-AGGGAGCACGAAAAACTGGT-3'	NM_011613.3
RANKL	Reverse	5'-GGAAGGGTTGGACACCTGAA-3'	
RUNX2	Forward	5'-GCCTCCAGCACCCTATACCC-3'	NM_001271631.1
RUNX2	Reverse	5'-CACATAGGTCCCCATCTGCC-3'	

SGK1	Forward	5'-GGGTGCCAAGGATGACTTTA-3'	NM_001161845.2
SGK1	Reverse	5'-CTCGGTAAACTCGGGATCAA-3'	
YwHAZ	Forward	5'-GGGGTGATTGGCAAAAGGTA-3'	NM_001356569.1
YwHAZ	Reverse	5'-CGACTTGGAAGCACAGAACT-3'	

Table 2: Table of antibodies.

Antibody	Manufacturer	Product Number
β-actin	Cell Signaling Technology	4967
GR	Santa Cruz Biotechnology	sc-1004
Osteocalcin	abcam	ab93876
RANKL	R&D Systems	AF462
RUNX2	abcam	ab76956

References

Alberta Energy Regulator. (2015). Alberta's energy reserves 2014 and

supply/demand outlook 2015-2024: Calgary. Alberta, Canada, Alberta

Energy Regulator Statistical Series ST98-2015.

Bartlett, A. J., Frank, R. A., Gillis, P. L., Parrott, J. L., Marentette, J. R., Brown,

L. R., Hooey, T., Vanderveen, R., McInnis, R., Brunswick, P., Shang, D.,

Headley, J. V., Peru, K. M., & Hewitt, L. M. (2017). Toxicity of

naphthenic acids to invertebrates: Extracts from oil sands process-affected

water versus commercial mixtures. Environmental Pollution (Barking,

Essex: 1987), 227, 271-279. https://doi.org/10.1016/j.envpol.2017.04.056

Brassey, C. A., Gardiner, J. D., & Kitchener, A. C. (2018). Testing hypotheses for the function of the carnivoran baculum using finite-element analysis. Proceedings of the Royal Society B: Biological Sciences, 285(1887), 20181473. https://doi.org/10.1098/rspb.2018.1473

Briot, K., & Roux, C. (2015). Glucocorticoid-induced osteoporosis. *RMD Open*, *1*(1). https://doi.org/10.1136/rmdopen-2014-000014

Chen, S.-M., Peng, Y.-J., Wang, C.-C., Su, S.-L., Salter, D. M., & Lee, H.-S.
(2018). Dexamethasone Down-regulates Osteocalcin in Bone Cells
through Leptin Pathway. *International Journal of Medical Sciences*, 15(5), 507–516. https://doi.org/10.7150/ijms.21881

- Clemente, J. S., & Fedorak, P. M. (2005). A review of the occurrence, analyses, toxicity, and biodegradation of naphthenic acids. *Chemosphere*, 60(5), 585–600. https://doi.org/10.1016/j.chemosphere.2005.02.065
- Danion, M., Deschamps, M.-H., Thomas-Guyon, H., Bado-Nilles, A., Le Floch,
 S., Quentel, C., & Sire, J.-Y. (2011). Effect of an experimental oil spill on vertebral bone tissue quality in European sea bass (Dicentrarchus labrax
 L.). *Ecotoxicology and Environmental Safety*, 74(7), 1888–1895.
 https://doi.org/10.1016/j.ecoenv.2011.07.027
- Frank, R. A., Kavanagh, R., Burnison, B. K., Headley, J. V., Peru, K. M., Der Kraak, G. V., & Solomon, K. R. (2006). Diethylaminoethyl-cellulose clean-up of a large volume naphthenic acid extract. *Chemosphere*, 64(8), 1346–1352. https://doi.org/10.1016/j.chemosphere.2005.12.035
- Garcia-Garcia, E., Ge, J. Q., Oladiran, A., Montgomery, B., El-Din, M. G., Perez-Estrada, L. C., Stafford, J. L., Martin, J. W., & Belosevic, M. (2011).

Ozone treatment ameliorates oil sands process water toxicity to the mammalian immune system. *Water Research*, *45*(18), 5849–5857. https://doi.org/10.1016/j.watres.2011.08.032

Hauge, E. M., Qvesel, D., Eriksen, E. F., Mosekilde, L., & Melsen, F. (2001).
Cancellous bone remodeling occurs in specialized compartments lined by cells expressing osteoblastic markers. *Journal of Bone and Mineral Research: The Official Journal of the American Society for Bone and Mineral Research*, *16*(9), 1575–1582.
https://doi.org/10.1359/jbmr.2001.16.9.1575

- He, Y., Zhang, Y., Martin, J. W., Alessi, D. S., Giesy, J. P., & Goss, G. G. (2018). In vitro assessment of endocrine disrupting potential of organic fractions extracted from hydraulic fracturing flowback and produced water (HF-FPW). *Environment International*, *121*, 824–831. https://doi.org/10.1016/j.envint.2018.10.014
- Kavanagh, R. J., Frank, R. A., Solomon, K. R., & Van Der Kraak, G. (2013).
 Reproductive and health assessment of fathead minnows (Pimephales promelas) inhabiting a pond containing oil sands process-affected water. *Aquatic Toxicology (Amsterdam, Netherlands), 130–131, 201–209.*https://doi.org/10.1016/j.aquatox.2013.01.007
- Knag, A. C., Verhaegen, S., Ropstad, E., Mayer, I., & Meier, S. (2013). Effects of polar oil related hydrocarbons on steroidogenesis in vitro in H295R cells.

Chemosphere, 92(1), 106–115.

https://doi.org/10.1016/j.chemosphere.2013.02.046

- Komori, T. (2016). Cell Death in Chondrocytes, Osteoblasts, and Osteocytes. International Journal of Molecular Sciences, 17(12). https://doi.org/10.3390/ijms17122045
- Kuo, T., Lew, M. J., Mayba, O., Harris, C. A., Speed, T. P., & Wang, J.-C.
 (2012). Genome-wide analysis of glucocorticoid receptor-binding sites in myotubes identifies gene networks modulating insulin signaling. *Proceedings of the National Academy of Sciences*, 109(28), 11160–11165.
 https://doi.org/10.1073/pnas.1111334109
- Lee, J.-M., Kim, M.-G., Byun, J.-H., Kim, G.-C., Ro, J.-H., Hwang, D.-S., Choi, B.-B., Park, G.-C., & Kim, U.-K. (2017). The effect of biomechanical stimulation on osteoblast differentiation of human jaw periosteum-derived stem cells. *Maxillofacial Plastic and Reconstructive Surgery*, 39(1), 7. https://doi.org/10.1186/s40902-017-0104-6
- Li, C., Fu, L., Lillico, D. M. E., Belosevic, M., Stafford, J. L., & Gamal El-Din, M. (2019). Exposure to Organic Fraction Extracted from Oil Sands Process-Affected Water Has Negligible Impact on Pregnancy and Lactation of Mice. *Environmental Science & Technology*, *53*(12), 7083– 7094. https://doi.org/10.1021/acs.est.9b01965
- Li, C., Fu, L., Stafford, J., Belosevic, M., & Gamal El-Din, M. (2017). The toxicity of oil sands process-affected water (OSPW): A critical review.

Science of The Total Environment, *601–602*, 1785–1802. https://doi.org/10.1016/j.scitotenv.2017.06.024

Li, H., Li, T., Fan, J., Li, T., Fan, L., Wang, S., Weng, X., Han, Q., & Zhao, R. C. (2015). MiR-216a rescues dexamethasone suppression of osteogenesis, promotes osteoblast differentiation and enhances bone formation, by regulating c-Cbl-mediated PI3K/AKT pathway. *Cell Death & Differentiation*, 22(12), 1935–1945. https://doi.org/10.1038/cdd.2015.99

Lister, A., Nero, V., Farwell, A., Dixon, D. G., & Van Der Kraak, G. (2008).
Reproductive and stress hormone levels in goldfish (Carassius auratus) exposed to oil sands process-affected water. *Aquatic Toxicology* (*Amsterdam, Netherlands*), 87(3), 170–177.
https://doi.org/10.1016/j.aquatox.2008.01.017

Livak, K. J., & Schmittgen, T. D. (2001). Analysis of relative gene expression data using real-time quantitative PCR and the 2(-Delta Delta C(T))
Method. *Methods (San Diego, Calif.)*, 25(4), 402–408.
https://doi.org/10.1006/meth.2001.1262

Loughery, J. R., Marentette, J. R., Frank, R. A., Hewitt, L. M., Parrott, J. L., & Martyniuk, C. J. (2019). Transcriptome Profiling in Larval Fathead
Minnow Exposed to Commercial Naphthenic Acids and Extracts from Fresh and Aged Oil Sands Process-Affected Water. *Environmental Science & Technology*, *53*(17), 10435–10444.
https://doi.org/10.1021/acs.est.9b01493

- Mahaffey, A., & Dubé, M. (2016). Review of the composition and toxicity of oil sands process-affected water. *Environmental Reviews*, 25(1), 97–114. https://doi.org/10.1139/er-2015-0060
- Marentette, J. R., Frank, R. A., Bartlett, A. J., Gillis, P. L., Hewitt, L. M., Peru, K. M., Headley, J. V., Brunswick, P., Shang, D., & Parrott, J. L. (2015).
 Toxicity of naphthenic acid fraction components extracted from fresh and aged oil sands process-affected waters, and commercial naphthenic acid mixtures, to fathead minnow (Pimephales promelas) embryos. *Aquatic Toxicology*, *164*, 108–117. https://doi.org/10.1016/j.aquatox.2015.04.024
- Marentette, J. R., Sarty, K., Cowie, A. M., Frank, R. A., Hewitt, L. M., Parrott, J. L., & Martyniuk, C. J. (2017). Molecular responses of Walleye (Sander vitreus) embryos to naphthenic acid fraction components extracted from fresh oil sands process-affected water. *Aquatic Toxicology*, *182*, 11–19. https://doi.org/10.1016/j.aquatox.2016.11.003
- McClung, M. R., Grauer, A., Boonen, S., Bolognese, M. A., Brown, J. P., Diez-Perez, A., Langdahl, B. L., Reginster, J.-Y., Zanchetta, J. R., Wasserman, S. M., Katz, L., Maddox, J., Yang, Y.-C., Libanati, C., & Bone, H. G. (2014). Romosozumab in postmenopausal women with low bone mineral density. *The New England Journal of Medicine*, *370*(5), 412–420. https://doi.org/10.1056/NEJMoa1305224
- Philibert, D. A., Lyons, D. D., Qin, R., Huang, R., El-Din, M. G., & Tierney, K.B. (2019). Persistent and transgenerational effects of raw and ozonated oil

sands process-affected water exposure on a model vertebrate, the zebrafish. *Science of The Total Environment*, *693*, 133611. https://doi.org/10.1016/j.scitotenv.2019.133611

- Plotkin, L. I., & Bellido, T. (2016). Osteocytic signalling pathways as therapeutic targets for bone fragility. *Nature Reviews. Endocrinology*, *12*(10), 593–605. https://doi.org/10.1038/nrendo.2016.71
- Rauch, A., Seitz, S., Baschant, U., Schilling, A. F., Illing, A., Stride, B., Kirilov, M., Mandic, V., Takacz, A., Schmidt-Ullrich, R., Ostermay, S., Schinke, T., Spanbroek, R., Zaiss, M. M., Angel, P. E., Lerner, U. H., David, J.-P., Reichardt, H. M., Amling, M., ... Tuckermann, J. P. (2010).
 Glucocorticoids Suppress Bone Formation by Attenuating Osteoblast Differentiation via the Monomeric Glucocorticoid Receptor. *Cell Metabolism*, *11*(6), 517–531. https://doi.org/10.1016/j.cmet.2010.05.005
- Rodríguez-Estival, J., & Smits, J. E. G. (2016). Small mammals as sentinels of oil sands related contaminants and health effects in northeastern Alberta,
 Canada. *Ecotoxicology and Environmental Safety*, *124*, 285–295.
 https://doi.org/10.1016/j.ecoenv.2015.11.001

Rogers, V. V., Liber, K., & MacKinnon, M. D. (2002). Isolation and characterization of naphthenic acids from Athabasca oil sands tailings pond water. *Chemosphere*, 48(5), 519–527. https://doi.org/10.1016/S0045-6535(02)00133-9

- Seeman, E. (2003). Periosteal bone formation—A neglected determinant of bone strength. *The New England Journal of Medicine*, 349(4), 320–323. https://doi.org/10.1056/NEJMp038101
- Silva, B. C., & Bilezikian, J. P. (2015). Parathyroid hormone: Anabolic and catabolic actions on the skeleton. *Current Opinion in Pharmacology*, 22, 41–50. https://doi.org/10.1016/j.coph.2015.03.005
- Sonne, C., Dyck, M., Rigét, F. F., Beck Jensen, J.-E., Hyldstrup, L., Letcher, R. J., Gustavson, K., Gilbert, M. T. P., & Dietz, R. (2015). Penile density and globally used chemicals in Canadian and Greenland polar bears. *Environmental Research*, 137, 287–291.
 https://doi.org/10.1016/j.envres.2014.12.026
- Stein, E. M., Silva, B. C., Boutroy, S., Zhou, B., Wang, J., Udesky, J., Zhang, C., McMahon, D. J., Romano, M., Dworakowski, E., Costa, A. G., Cusano, N., Irani, D., Cremers, S., Shane, E., Guo, X. E., & Bilezikian, J. P. (2013). Primary Hyperparathyroidism is Associated with Abnormal Cortical and Trabecular Microstructure and Reduced Bone Stiffness in Postmenopausal Women. *Journal of Bone and Mineral Research : The Official Journal of the American Society for Bone and Mineral Research*, 28(5), 1029–1040. https://doi.org/10.1002/jbmr.1841
- Stockley, P., Ramm, S. A., Sherborne, A. L., Thom, M. D. F., Paterson, S., & Hurst, J. L. (2013). Baculum morphology predicts reproductive success of

male house mice under sexual selection. *BMC Biology*, *11*(1), 66. https://doi.org/10.1186/1741-7007-11-66

Thomas, P. J., Newell, E. E., Eccles, K., Holloway, A. C., Idowu, I., Xia, Z., Hassan, E., Tomy, G., & Quenneville, C. (2021). Co-exposures to trace elements and polycyclic aromatic compounds (PACs) impacts North American river otter (Lontra canadensis) baculum. *Chemosphere*, 265, 128920. https://doi.org/10.1016/j.chemosphere.2020.128920

Vander Meulen, I. J., Schock, D. M., Parrott, J. L., Mundy, L. J., Pauli, B. D., Peru, K. M., McMartin, D. W., & Headley, J. V. (2021). Characterization of naphthenic acid fraction compounds in water from Athabasca oil sands wetlands by Orbitrap high-resolution mass spectrometry. *Science of The Total Environment*, 780, 146342.

https://doi.org/10.1016/j.scitotenv.2021.146342

Wang, D., Christensen, K., Chawla, K., Xiao, G., Krebsbach, P. H., & Franceschi, R. T. (1999). Isolation and Characterization of MC3T3-E1 Preosteoblast Subclones with Distinct In Vitro and In Vivo Differentiation/Mineralization Potential. *Journal of Bone and Mineral Research*, *14*(6), 893–903. https://doi.org/10.1359/jbmr.1999.14.6.893

Weinstein, R. S., Jilka, R. L., Parfitt, A. M., & Manolagas, S. C. (1998).Inhibition of osteoblastogenesis and promotion of apoptosis of osteoblasts and osteocytes by glucocorticoids. Potential mechanisms of their deleterious effects on bone. *Journal of Clinical Investigation*, *102*(2), 274–282.

- Wu, C., De Visscher, A., & Gates, I. D. (2019). On naphthenic acids removal from crude oil and oil sands process-affected water. *Fuel*, 253, 1229– 1246. https://doi.org/10.1016/j.fuel.2019.05.091
- Xu, S., Guo, R., Li, P., Li, K., Yan, Y., Chen, J., Wang, G., Brand-Saberi, B.,
 Yang, X., & Cheng, X. (2019). Dexamethasone interferes with osteoblasts
 formation during osteogenesis through altering IGF-1-mediated
 angiogenesis. *Journal of Cellular Physiology*, 234(9), 15167–15181.
 https://doi.org/10.1002/jcp.28157

3. Chapter 3

Prepared for publication.

3.1. Preface and significance to thesis

Our lab had previously found that commercial naphthenic acids (NAs) were PPAR γ ligands (*Jamshed and Holloway, unpublished data*), and the induction of PPAR γ activity by environmental contaminants has previously been demonstrated to induce adipogenesis in adipocyte progenitor cells (Boucher et al., 2014). In addition, OSPW (which contains naphthenic acids) had previously been found to contain PPAR γ ligands and induce adipogenesis in adipocyte progenitor cells (Peng et al., 2016), Dr. Richard Frank and Dr. Mark Hewitt at Environment and Climate Change Canada generously donated a sample of NAFC of OSPW from oil sands tailings water in the AOSR which had not been assessed for its effects on adipogenesis. We found that although commercial NAs were PPAR γ ligands, they did not induce adipogenesis, nor did NAFC. Interestingly, we found that the highest dose of NAFC inhibited the expression of gene markers of PPAR γ activation *Fabp4*, *Cd36* and *C/ebp-β* in mature adipocytes, suggesting a potential inhibition of PPAR γ by NAFC.

3.2. Author Contribution

RMG, PJT, LJ and ACH were responsible for the experimental design of this study. LMH and RAF provided the NAFC stock solution. RMG was the primary contributor to the data in Figure 2, Figure 3, Figure 4, and Figure 4. LJ was the primary contributor to the data in Figure 1 and she and Raji Rajilingham contributed to preliminary lab experiments. RMG wrote the first draft of the manuscript and the rest of the authors provided edits and feedback on the manuscript.

3.3. Naphthenic acid fraction components from oil sands process affected water from Athabasca Oil Sands Region do not induce adipogenesis in 3T3-L1 cells

Naphthenic acid fraction components from oil sands process affected water from Athabasca Oil Sands Region do not induce adipogenesis in 3T3-L1 cells

Robert M. Gutgesell¹, Laiba Jamshed¹, Raji Rajilingham¹, Richard A. Frank², L. Mark Hewitt², Philippe J. Thomas³ Alison C. Holloway¹

- 1. Department of Obstetrics and Gynecology, McMaster University, Hamilton, ON, Canada
- 2. Water Science and Technology Directorate, Environment and Climate Change Canada, Burlington, ON, Canada
- 3. Environment and Climate Change Canada, National Wildlife Research Centre, Ottawa ON., Canada

Address for correspondence:	Dr. Alison Holloway	
-	Department of Obstetrics & Gynecology	
	McMaster University	
	RM HSC-3N52	
	1280 Main Street West	
	Hamilton, Ontario, Canada, L8S 4K1	
	Phone: (905) 525-9140 ext. 22130	
	E-mail: hollow@mcmaster.ca	

Short title: Napthenic Acid Fraction Componenents Do Not Induce Adipogenesis

Key words: Naphthenic acids, oil sands process affected water, OSPW, adipocyte, adipogenesis, lipid metabolism, PPAR gamma

<u>Abstract</u>

The Athabasca Oil Sands Region (AOSR) contain naturally occurring bitumen which is extracted from sand, clay and water through oil sands mining processes. Bitumen contains several potential toxicants including naphthenic acid fraction components (NAFCs) that are solubilized and concentrated in oil sands process affected water (OSPW) during bitumen extraction processes. OSPW is then stored within the AOSR in tailings ponds. NAFCs are considered among the most toxic components in OSPW. OSPW has been reported to induce adipogenesis in adipocytes an effect that has been reported to be mediated via peroxisome proliferator-activated-receptor γ (PPAR γ)-mediated signaling. However, the effects of NAFCs on adipogenesis and lipid homeostasis in mature adipocytes is unknown. Therefore, the aims of this study were to 1) evaluate the effect of NAFCs on adipogenesis and lipid metabolism in 3T3-L1 murine adipocyte progenitor cells and 2) determine the effects of NAFCs on the PPAR γ signaling pathway. Exposure to environmentally relevant concentrations of NAFCs (1.25-125mg/L) did not affect the transcriptional regulation of adipogenesis, markers of mature adipocytes or lipid content during differentiation. Mature adjocytes treated with NAFCs had decreased expression of markers of PPARy signalling but this did not result in any changes in cellular lipid content. These data indicate that NAFCs do not induce adipogenesis but may inhibit PPARy signaling in mature adipocytes. Our data suggests that NAFCs may impair lipid metabolism in adipocytes in chronically exposed animals. Taken

together these data suggest that NAFC exposure may affect lipid homeostasis via direct effects on adipose tissue.
Introduction

Oil and gas extraction in the Athabasca Oil Sands Region (AOSR) accounts for 21% of Alberta's gross domestic product (Government of Canada, 2021). The extraction of bitumen from the oil sands uses large amounts of water (oil sands process-affected water; OSPW) which is used and then stored within the AOSR in tailings ponds (Wu et al., 2019). Tailings ponds are separated from natural waterways and comprise an area in the AOSR of approximately 176km² containing 766x10⁶ m³ of tailings in 2011 (Small et al., 2015). The class of chemicals in OSPW that are considered to have the most toxic effects are the naphthenic acid fraction components (NAFCs). NAFCs are composed of the carboxylic acids and acid-extractable organic compounds of bitumen (Frank et al., 2006; Headley et al., 2011). The effects of exposure to naphthenic acids (NAs) have been widely studied in a variety of invertebrate and vertebrate species (Bartlett et al., 2017; Clemente & Fedorak, 2005; Marentette et al., 2015). However, many of these studies have used commercially available technical NA mixtures as a surrogate for the NAFCs found in OSPW (Bartlett et al., 2017; Brown & Ulrich, 2015). Importantly, technical NA mixtures are less complex with fewer ring structures and oxygen species than NAFCs from OSPW and may have different toxicities (Marentette et al., 2015).

There is increasing concern that exposure to chemicals from oil and gas extraction may be adversely affecting the health of wildlife and human populations living in close proximity to active extraction sites. There is evidence

that many environmental compounds with endocrine disrupting properties (EDCs) have the capability to induce/promote adipogenesis and thus contribute to the development of obesity (Biemann et al., 2021; Gutgesell et al., 2020; Mallhi et al., 2021). Normal adipose tissue functioning is required to adapt to the seasonal variations in food supply, support milk production, and provide endocrine signaling that contributes to fertility, immune function and inflammatory responses (McNamara & Huber, 2018; Wensveen et al., 2015; Young, 1976; Zwick et al., 2018). White adipose tissue (WAT) adapts to energetic stress by increasing the number of adipocytes (hyperplasia), increasing the size of adipocytes (hypertrophy) or involution. Impaired adipose tissue adaptation is associated with chronic inflammation of WAT, accumulation of extracellular matrix and dysregulation of mitochondrial biogenesis, and altered endocrine function of adipocytes (Chouchani & Kajimura, 2019). These attributes of impaired adipose tissue adaptation are associated with a variety of metabolic disorders (Chouchani & Kajimura, 2019). Healthy adipose tissue remodeling requires a balance of hypertrophy and hyperplasia, requiring regulated differentiation of pre-adipocytes into mature adipocytes. Dysregulation of adipocyte proliferation and differentiation is associated with metabolic disease (Muir et al., 2016). There are two populations of adipocyte progenitor cells within the stromal vascular fraction of adipose tissue, mesenchymal stem cells (MSCs) and preadipocytes (Cawthorn et al., 2012). Adipogenesis requires path selection of MSCs to preadipocytes and then differentiation from preadipocytes to mature adipocytes. Adipocyte differentiation from preadipocytes requires tightly regulated expression of transcription factors including CCAAT/enhancer-binding protein family (C/EBP), sterol regulatory element binding protein-1 (SREBP-1) and peroxisome proliferator-activator receptor γ (PPAR γ) (Arimochi et al., 2016).

Many EDCs that have been shown to impact adipogenesis exert their effects via PPAR γ signaling. PPAR γ is a nuclear receptor and transcription factor central to regulating adipogenesis (Tontonoz et al., 1994). PPAR γ agonists induce the transcriptional activation of adipogenesis by interacting with co-activators and co-repressors like the CCAAT/enhancer-binding proteins (C/EBPs) and Krüppellike factors (KLF)(Rosen & MacDougald, 2006). In addition to its critical role in differentiating adipocytes, PPAR γ is also required for normal functioning of adipocyte metabolism and the maintenance of mature adipocytes (Spiegelman, 1997). In mature adipocytes the activation of PPAR γ promotes a net flux of fatty acids from circulation into adipocytes and increased insulin sensitivity in adipocytes (Lehrke & Lazar, 2005). Conversely, the inhibition of PPAR γ activity in mature adipocytes resulted in a decrease in cell size and triglyceride content and an increase in lipolysis and an increase in the rate of free fatty acid uptake (Tamori et al., 2002).

Several environmental contaminants including bisphenol A, tributyltin, triphenyl phosphate, and triflumizole have been found to induce adipogenesis by activating PPARy, (González-Casanova et al., 2020). Previous studies have shown that OSPW contains PPARy ligands and can induce lipid accumulation in

62

3T3-L1 pre-adipocytes (Peng et al., 2016). Peng et al. (2016) also reported that while the nonpolar fraction of OSPW that should contain most NAFCs did contain PPARγ ligands. However, treatment with a single concentration of this fraction of OSPW did not induce adipogenesis in 3T3-L1 pre-adipocytes in the presence of factors inducing adipogenesis. In this study we expand on this previous work by examining the effects of an environmentally relevant range of doses of isolated NAFCs from the AOSR (Frank et al., 2006) on both 3T3-L1 preadipocytes during differentiation in the absence of other factors inducing adipogenesis, and on lipid metabolism of mature adipocytes.

Materials & methods

Naphthenic Acids

Naphthenic acid extracts (NAFC) were isolated from tailings pond water collected from Syncrude Canada Ltd. West In-pit settling basin in Fort McMurray, Alberta, Canada in 2009, using a procedure described in (Frank et al., 2006). A stock solution was prepared in 0.05 M NaOH with a final concentration of 2504 mg/L determined via liquid chromatography/quadrupole mass spectrometry with time-of-flight detection (LC/QTofF) as previously described in Marentette et al., (2017).

MC 3T3-L1 cell culture & treatment

3T3-L1 preadipocytes for adipogenesis assay (AD)

MC 3T3-L1 mouse embryonic fibroblasts were maintained in Dulbecco's modified Eagle's medium (DMEM) containing 1g/L glucose (Corning, Corning, NY, USA), supplemented with 10% fetal bovine serum (Hyclone, Logan, UT, USA), 1% L-glutamine (Gibco, Waltham, MA, USA), and 1% penicillin/streptomycin (P/S) (Gibco). At 70% confluence, cells were passaged to 6-well plates and left to reach confluence. We exposed 3T3-L1 preadipocytes to NAFCs in the absence of insulin, IBMX and dexamethasone; these factors are known to cause differentiation of 3T3-L1 cells from preadipocytes into mature adipocytes (Zebisch et al., 2012). Two days after 100% confluence (day 0), 3T3-L1 preadipocytes were treated with vehicle (control), 1.25-125mg/L NAFC,

rosiglitazone (2µM, Sigma), and differentiation media. The dose range covers the concentrations of NAFCs that have been reported in surface waters, wetlands and tailings ponds in the AOSR and is in line with other studies investigating the effects of NAFCs have used (Bartlett et al., 2017; Vander Meulen et al., 2021). Two positive control groups were used to compare NAFC treated cells to differentiated cells and cells treated with a PPAR γ agonist. First, to compare the effects of NAFC on adipogenesis to a known differentiation stimulus, one group of cells was treated using the differentiation protocol described below and were harvested on day 8. Second, rosiglitazone (2µM, Sigma), a PPAR γ agonist, was added to the media to assess PPAR γ activation. Treatment media was replaced every 2 days until the cells were fully differentiated into mature adipocytes (i.e. 8 days).

Mature 3T3-L1 adipocytes

In experiments investigating the effects of commercial NA and NAFC in mature adipocytes, 3T3-L1 cells were differentiated using the following differentiation protocol: 2 days after cells had reached 100% confluence (day 0) cells received differentiation media (DM) (DMEM containing 4.5 g/L glucose, supplemented with 0.5 mM IBMX, 2µg/mL insulin, 2µM rosiglitazone and 0.25 µM dexamethasone). On day 3 differentiation media was removed and replaced with post differentiation media (PDM) (DMEM containing 4.5 g/L glucose and supplemented with 2µg/mL insulin). PDM was removed and replaced every 2 days. On day 9 differentiated cells were treated with vehicle (control), 1.25-125mg/L naphthenic acids extracted (NAFC) or Rosiglitazone (2μM, Sigma), in PDM for 48 hours. The pH of the media was adjusted to be similar across all groups using NaOH or HCl after the addition of NAFC.

Cell viability

Cells were seeded and treated in 96-well plates following the protocols described above. For the DA cells, after 8 days of treatment in cells treated in the absence of other differentiation factors 10% alamarBlue reagent (BioRad Laboratories, Hercules, CA, USA) was added to fresh media and cells were incubated for 2h. For the mature adipocytes, after 48h of treatment, 10% alamarBlue reagent (BioRad Laboratories, Hercules, CA, USA) was added to fresh media and cells were incubated for 2h. Absorbance was then read at 570nm in a plate reader (Synergy H1 microplate reader, Norgen BioTek, Thorold, ON, Canada) with a reference read at 600nm. Compounds were deemed to be cytotoxic when viability was less than 80% of the control group.

PPAR-*γ* ligand assay

The PPARγ ligand activity of compounds was measured using a competitive binding fluorescence-polarization PPARγ Ligand Screening Assay Kit (Cayman Chemicals, Ann Arbor, MI, USA). The PPARγ binding activity was determined for a range of concentrations of commercial naphthenic acids (CNAs)

(0.025-250mg/L; Sigma) and rosiglitazone (0.035-112.9mg/L) according to the manufacturer's instructions. Fluorescence-polarization was measured using a plate reader (Norgen BioTek) with an excitation wavelength of 485nm and an emission wavelength of 528nm. Polarization values for each compound were presented in mP units. The half-maximal inhibitory concentration (IC₅₀) values were determined from a 4-parameter logistic regression. As the tested range of concentrations of CNA and NAFC did not maximally bind PPAR γ , it was assumed that 10,000mg/L of each maximally bound NAFC in the 4-parameter model.

Lipid staining and quantification

To assess neutral lipid content following differentiation in DA cells, after 8 days of treatment cells were washed with DPBS (Corning), fixed with 10% formalin and stained with Oil Red O (Sigma) for 20 min, then washed 3x with water as previously described (Ramírez-Zacarías et al., 1992). The Oil Red O dye was eluted with 60% 2-propanol, and absorbance was read at 510nm (Norgen BioTek). To assess lipid accumulation following NAFC treatment in mature adipocytes, after 48 hours of treatment with NAFC, Oil red O staining was assessed as described above.

Real-time quantitative PCR

For DA cells, after 8 days of treatment, cells were washed with phosphatebuffered saline (PBS) and RNA was extracted using TRIzol reagent (Invitrogen, Waltham, MA, USA). For the mature adipocytes, after 48h of treatment, cells were washed with PBS and RNA was extracted using TRIzol reagent (Invitrogen). RNA concentrations were determined using the NanoDrop One Microvolume UV-Vis Spectrophotometer (Thermo Fisher Scientific, Waltham, MA, USA). Complementary DNA (cDNA) was made from 2 µg of mRNA using the High-Capacity cDNA Reverse Transcription Kit (Applied Biosystems, Waltham MA, USA) as per the manufacturer's instructions. Real-time quantitative PCR (RT-qPCR) was carried out using PerfeCTa SYBR Green FastMix (Quanta Biosciences, Gaithersburg, MD, USA) on the CFX384 Touch Real Time PCR Detection System (Bio-Rad Laboratories). The PCR cycling settings included polymerase activation (95 °C for 10 m), followed by 40 cycles of denaturing (95 °C for 15s) and annealing/elongation (60 °C for 1m). Levels of gene expression were generated using the $2(-\Delta\Delta C(T))$ method (Livak & Schmittgen, 2001) and normalized using the geometric means of two reference genes: Tyrosine 3-monooxygenase/tryptophan 5-monooxygnase activation protein zeta (Ywhaz) and peptidylprolyl isomerase A (Ppia). Gene targets for adipocyte differentiation: peroxisome proliferator receptor activator gamma (*Ppary*), CCAAT enhancer binding protein beta ($C/EBP\beta$), and cluster of differentiation 36 (CD36), were measured. A marker of mature adipocytes, fatty acid binding protein 4 (Fabp4), was also measured. Primer sequences are in Table 1.

Statistical analysis

Data are expressed as the means +/- standard errors of the means (SEM). Statistical analyses were carried out using GraphPad Prism 9.2 (GraphPad Software Inc, San Diego, CA, USA). One-way analysis of variance (ANOVA) followed by post-hoc testing (Dunnet multiple comparisons) when significance was indicated (p<0.05).

Results

PPARy ligand activity of naphthenic acids

To determine whether there is an interaction between PPAR γ , CNA, and NAFC, we performed a fluorescence-polarization based competitive binding assay using cell-free human PPAR γ . CNA competitively inhibited the binding of the PPAR γ -flourescent probe (FP) to PPAR γ (Fig. 1). The half maximal concentration (IC₅₀) values for rosiglitazone were 2.41mg/L (8.61 μ M), 82.0mg/L for CNA and 269 mg/L for NAFC.

Effect of NAFC on 3T3-E1 cell viability

3T3-L1 cells treated with 125 mg/L NAFC for 8 days did not meet the threshold of 80% of the viability of cells treated with control media and thus doesn't meet the threshold for cytotoxicity. Treatment with 1.25 and 12.5 mg/L of NAFC had no effect on cell viability after 8 days of treatment (Fig. 2a).

NAFC induce adipogenesis in 3T3-L1 pre-adipocytes

To investigate whether NAFCs can induce adipogenesis, we determined lipid accumulation in 3T3-L1 cells treated with 1.25-125mg/L NAFC. There was no significant change in lipid accumulation at any dose of NAFC after 8 days of treatment in the absence of other factors inducing differentiation (Fig. 2b). Both the rosiglitazone treated cells and the cells treated with pro-differentiation compounds had increased lipid content (Fig. 2b).

To further evaluate the effect that NAFC have on PPAR γ activity we measured the expression of genes that are transcriptionally regulated by PPAR γ

activation namely *Ppary*, *C/ebpβ* and *Cd36* as well as the marker of mature adipocytes *Fabp4* (Fig. 3). There was no significant change in *Ppary* or *Cd36* expression at any dose of NAFC (Fig. 3 a, c, e, h), while rosiglitazone treated cells and cells treated with IBMX, insulin and dexamethasone (differentiation media) both had higher *Cd36* expression. 125 mg/L of NAFC inhibited the expression of *C/ebpβ* while cells treated with both rosiglitazone and differentiation media had higher expression of *C/ebpβ* compared to cells that were treated with media in the absence of any differentiation stimulus (Fig. 3 c, g). All other doses of NAFC had no effect on *C/ebpβ* expression. 125mg/L NAFC also induced increased expression of *Fabp4*, although to a lesser magnitude than as the rosiglitazone or differentiation media positive controls (p<0.0001, Fig. 3b).

Effect of NAFC on mature 3T3-L1 cell viability

None of the treatment doses of NAFC had any significant effect on cell viability (Fig. 4b).

NAFC inhibits markers of lipid homeostasis in differentiated 3T3-L1 adipocytes

To evaluate whether treatment with NAFC enhanced lipid accumulation in mature adipocytes, we differentiated 3T3-L1 cells and then treated them for 48 hours with NAFC. 1.25mg/L NAFC-treated cells had higher lipid content than control cells but this increase in lipid content was not observed at any other concentration tested (Fig. 4a,g). Next, we measured the gene markers of PPAR γ activity *Ppar\gamma*, *C/ebp\beta* and *Cd36* as well as the marker of mature adipocytes

Fabp4 in NAFC treated mature adipocytes. We found that while there was no change in *Ppary* expression for any groups, cells treated with 125mg/L of NAFC had significantly lower expression of *C/ebpβ*, *Cd36*, and *Fabp4* (Fig. 4c-f). Cells treated with both rosiglitazone and 25mg/L NAFC also had significantly lower *C/ebpβ* expression than controls (Fig. 4c).

Discussion

EDCs have been reported to affect adipose tissue development and function (Egusquiza & Blumberg, 2020; González-Casanova et al., 2020). There is evidence that OSPW and NAFC can perturb endocrine homeostasis, the effects of NAFC exposure on adipogenesis and the function of mature adipocytes remains to be elucidated (Leclair et al., 2015; Li et al., 2017; Wiseman et al., 2013). In this study we exposed 3T3-L1 cells to doses of NAFC from 1.25 mg/L to 125mg/L which encompasses the range of NAFCs reported in reference surface water sources, industrially impacted wetlands and tailings ponds (Vander Meulen et al., 2021). There is clear evidence in the literature that exposure to environmental toxicants that act as PPARγ ligands can increase adipogenesis and alter adipose tissue function (Kirk et al., 2021; Riu et al., 2011). Although both CNAs and NAFCs act as PPARγ ligands, it appears that NAFC does not bind PPARγ as strongly (Fig. 1). NAFC exposure did not cause increased lipid accumulation in 3T3-L1 cells; an outcome that is typical of exposure to PPARy agonists in this cell line (Zhang et al., 1996). PPARy activation alone is enough to induce differentiation (Tontonoz et al., 1994). Similarly, we observed an induction of differentiation and lipid accumulation in cells treated with the PPARy agonist rosiglitazone in the absence of other differentiation factors (Figs. 2a, 3). The lack of a similar effect in the NAFC treated cells argues against this mixture containing any significant PPARy binding activity. Conversely, OSPW and the wastewater from unconventional oil and gas extraction wastewater did induce adipogenesis in 3T3-L1 preadipocytes (Kassotis et al., 2018; Peng et al., 2016). However, our findings that NAFCs did not induce adipogenesis are in line with what Peng et al. (2016) found where the non-polar fraction of OSPW containing most NAs did induce PPARy activity in a reporter assay but did not increase lipid accumulation in adjocytes. Taken together, these data suggest that while some compounds related to oil and gas extraction can induce adipogenesis, NAFCs are unlikely to be the primary driver of this effect.

Although NAFCs did not have a significant effect on adipogenesis, there is some evidence that high doses of NAFC inhibited the transcription of PPAR γ regulated gene targets in mature adipocytes, suggesting that NAFCs may act as inhibitors of PPAR γ activity in existing adipose tissue. Mature adipocytes treated for 48h with 125mg/L of NAFC had reduced expression of *Fabp4*, and *Cd36*, and treatment with both 25 mg/L and 125 mg/L of NAFC inhibited the expression of *C/ebp-β* (Fig. 4d-f). C/ebp-β is part of the group of transcription factors that, along with PPAR γ and the other C/EBP family members, is responsible for the expression of adipocyte markers in mature adipocytes (Rosen et al., 2000; Tamori et al., 2002). Inhibition of *Fabp4* and *C/ebp-\beta* expression is consistent with inhibition of PPAR γ activity. *Cd36* expression by NAFC in mature adipocytes may result in increased lipolysis and a dysfunction in the adipocytes ability to respond to metabolic stress, and to store and release energy. Peng et al. (2016) also reported lower *Fabp4* expression in 3T3-L1 cells treated with the non-polar fraction of OSPW. Our data here also suggests that NAFC from OSPW inhibit PPAR γ and may disrupt mature adipocyte function.

In summary, we have shown that exposure to NAFCs do not induce adipogenesis in 3T3-L1 preadipocytes but does inhibit PPARγ activity and the expression of genes required for adipocyte function in mature adipocytes. Therefore, these data suggest that exposure to NAFCs may have the potential to impair adipocyte function in chronically exposed animals which may impact their ability to adapt to energetic stresses and survive periods of low food availability.

Figures



Figure 1: CNA contains PPAR γ ligands

Polarization (mP) of competitive inhibition of PPAR γ -fluorescent probe (FP) binding to cell free PPAR γ . incubated with different concentrations of CNA and NAFC at concentrations from 0.025 mg/L to 250 mg/L and rosiglitazone (Rosi) 0.036-113 mg/L (0.1-316 μ M), in the presence of PPAR γ -FP. Each value is the mean +/- SEM (Rosi n=3, CNA n=2).







Figure 3: RNA markers of PPARy induced adipogenesis NAFC exposed adipocytes

(A-D) Expression of adipogenic marker genes *Ppary*, *Fabp4*, *C/ebp-\beta*, and *Cd36* in 3T3-L1 cells treated with 1.25-125 mg/L NAFC or 2µM Rosiglitazone or differentiation media for 8 days.

Each value is the mean +/- SEM (n=6). *p<0.05, ***p<0.001, ****p<0.0001, #p<0.05 positive control compared with the corresponding control group.





(A) Lipid accumulation of mature 3T3-L1 cells treated with 1.25-125 mg/L NAFC or 2μ M Rosiglitazone for 48 hours.

(B) Cell viability of mature 3T3-L1 cells treated with 1.25-125 mg/L NAFC or $2\mu M$ Rosiglitazone for 48 hours.

(C-F) Expression of adipogenic marker genes *Ppary*, *Fabp4*, *C/ebp-\beta*, and *Cd36* in 3T3-L1 cells treated with 1.25-125 mg/L NAFC or 2µM Rosiglitazone or for 48 hours.

Each value is the mean +/- SEM (n=6). *p<0.05, ***p<0.001, ****p<0.0001, #p<0.05 positive control compared with the corresponding control group.

Gene	Forward/	Sequence	Accession No.
	Reverse		
YwHAZ	Forward	5'-GGGGTGATTGGCAAAAGGTA-3'	NM_001356569.1
YwHAZ	Reverse	5'-CGACTTGGAAGCACAGAACT-3'	
PPIA	Forward	5'-GACAAAGTTCCAAAGACAGCAGAA-'3	NM_008907.2
PPIA	Reverse	5'-CCAAATCCTTTCTCTCCAGTGC-3'	
FABP4	Forward	5'-ATT TCC TTC AAA CTG GGC GTG-3'	NM_024406.3
FABP4	Reverse	5'-CTT TCC ATC CCA CTT CTG CAC-3'	
PPARy	Forward	5'-GCG GAA GAA GAG ACC TGG G-3'	NM_001127330.2
PPARy	Reverse	5'-GTG ACT TCT CCT CAG CCC G-3'	
CD36	Forward	5'-CTGTGTCCTTCTGAATCATTTAACC-3'	NM_001159558.1
CD36	Reverse	5'-TTCAGTCTCACACTGCTGCTAT-3'	
C/EBP-β	Forward	5'-CCG TGG TGG TTT CTC GA-3'	NM_001287739.1
C/EBP-β	Reverse	5'-TTT TTG CTC CCC CTA CTCC GG-3'	

Table 1: Primer sequences.

Funding source

This study was supported by a CIHR Project Grant (PJT 159651) to ACH.

Declaration of Competing Interest

None.

Acknowledgements

None.

Disclaimer

The views in this paper are only held by the authors and are not representative of the official policy of the authors' individual affiliations.

<u>References</u>

Arimochi, H., Sasaki, Y., Kitamura, A., & Yasutomo, K. (2016). Differentiation of preadipocytes and mature adipocytes requires PSMB8. *Scientific Reports*, 6(1), 26791. https://doi.org/10.1038/srep26791

Bartlett, A. J., Frank, R. A., Gillis, P. L., Parrott, J. L., Marentette, J. R., Brown,
L. R., Hooey, T., Vanderveen, R., McInnis, R., Brunswick, P., Shang, D.,
Headley, J. V., Peru, K. M., & Hewitt, L. M. (2017). Toxicity of
naphthenic acids to invertebrates: Extracts from oil sands process-affected
water versus commercial mixtures. *Environmental Pollution (Barking, Essex: 1987)*, 227, 271–279. https://doi.org/10.1016/j.envpol.2017.04.056

- Biemann, R., Blüher, M., & Isermann, B. (2021). Exposure to endocrinedisrupting compounds such as phthalates and bisphenol A is associated with an increased risk for obesity. *Best Practice & Research Clinical Endocrinology & Metabolism*, 35(5), 101546. https://doi.org/10.1016/j.beem.2021.101546
- Brown, L. D., & Ulrich, A. C. (2015). Oil sands naphthenic acids: A review of properties, measurement, and treatment. *Chemosphere*, *127*, 276–290. https://doi.org/10.1016/j.chemosphere.2015.02.003

Cawthorn, W. P., Scheller, E. L., & MacDougald, O. A. (2012). Adipose tissue stem cells meet preadipocyte commitment: Going back to the future. *Journal of Lipid Research*, 53(2), 227–246. https://doi.org/10.1194/jlr.R021089

- Chouchani, E. T., & Kajimura, S. (2019). Metabolic adaptation and maladaptation in adipose tissue. *Nature Metabolism*, 1(2), 189–200. https://doi.org/10.1038/s42255-018-0021-8
- Clemente, J. S., & Fedorak, P. M. (2005). A review of the occurrence, analyses, toxicity, and biodegradation of naphthenic acids. *Chemosphere*, 60(5), 585–600. https://doi.org/10.1016/j.chemosphere.2005.02.065
- Egusquiza, R., & Blumberg, B. (2020). Environmental Obesogens and Their Impact on Susceptibility to Obesity: New Mechanisms and Chemicals. *Endocrinology*, *161*. https://doi.org/10.1210/endocr/bqaa024
- Frank, R. A., Kavanagh, R., Burnison, B. K., Headley, J. V., Peru, K. M., Der Kraak, G. V., & Solomon, K. R. (2006). Diethylaminoethyl-cellulose clean-up of a large volume naphthenic acid extract. *Chemosphere*, 64(8), 1346–1352. https://doi.org/10.1016/j.chemosphere.2005.12.035
- González-Casanova, J. E., Pertuz-Cruz, S. L., Caicedo-Ortega, N. H., & Rojas-Gomez, D. M. (2020). Adipogenesis Regulation and Endocrine
 Disruptors: Emerging Insights in Obesity. *BioMed Research International*, 2020, e7453786. https://doi.org/10.1155/2020/7453786

Government of Canada, S. C. (2021, July 28). *The oil and gas sector in Canada: A year after the start of the pandemic*. https://www150.statcan.gc.ca/n1/pub/36-28-0001/2021007/article/00003-

- Gutgesell, R. M., Tsakiridis, E. E., Jamshed, S., Steinberg, G. R., & Holloway, A.
 C. (2020). Impact of pesticide exposure on adipose tissue development and function. *Biochemical Journal*, 477(14), 2639–2653. https://doi.org/10.1042/BCJ20200324
- Headley, J. V., Barrow, M. P., Peru, K. M., Fahlman, B., Frank, R. A., Bickerton,
 G., McMaster, M. E., Parrott, J., & Hewitt, L. M. (2011). Preliminary
 fingerprinting of Athabasca oil sands polar organics in environmental
 samples using electrospray ionization Fourier transform ion cyclotron
 resonance mass spectrometry. *Rapid Communications in Mass Spectrometry: RCM*, 25(13), 1899–1909. https://doi.org/10.1002/rcm.5062
- Kassotis, C. D., Nagel, S. C., & Stapleton, H. M. (2018). Unconventional oil and gas chemicals and wastewater-impacted water samples promote adipogenesis via PPARγ-dependent and independent mechanisms in 3T3–L1 cells. *The Science of the Total Environment*, 640–641, 1601–1610. https://doi.org/10.1016/j.scitotenv.2018.05.030
- Kirk, A. B., Michelsen-Correa, S., Rosen, C., Martin, C. F., & Blumberg, B.
 (2021). PFAS and Potential Adverse Effects on Bone and Adipose Tissue Through Interactions With PPARγ. *Endocrinology*, *162*(12), bqab194. https://doi.org/10.1210/endocr/bqab194
- Leclair, L. A., Pohler, L., Wiseman, S. B., He, Y., Arens, C. J., Giesy, J. P.,Scully, S., Wagner, B. D., van den Heuvel, M. R., & Hogan, N. S. (2015).In vitro assessment of endocrine disrupting potential of naphthenic Acid

fractions derived from oil sands-influenced water. *Environmental Science* & *Technology*, 49(9), 5743–5752. https://doi.org/10.1021/acs.est.5b00077

Lehrke, M., & Lazar, M. A. (2005). The Many Faces of PPARγ. *Cell*, *123*(6), 993–999. https://doi.org/10.1016/j.cell.2005.11.026

Li, C., Fu, L., Stafford, J., Belosevic, M., & Gamal El-Din, M. (2017). The toxicity of oil sands process-affected water (OSPW): A critical review. *Science of The Total Environment*, 601–602, 1785–1802. https://doi.org/10.1016/j.scitotenv.2017.06.024

Livak, K. J., & Schmittgen, T. D. (2001). Analysis of relative gene expression data using real-time quantitative PCR and the 2(-Delta Delta C(T))
Method. *Methods (San Diego, Calif.)*, 25(4), 402–408.
https://doi.org/10.1006/meth.2001.1262

Mallhi, T. H., Khokhar, A., Khan, Y. H., Alotaibi, N. H., & Khan, A. (2021).
Endocrine Disrupting Chemicals Induced Childhood Obesity. In M. S. H.
Akash, K. Rehman, & M. Z. Hashmi (Eds.), *Endocrine Disrupting Chemicals-induced Metabolic Disorders and Treatment Strategies* (pp. 147–160). Springer International Publishing. https://doi.org/10.1007/978-3-030-45923-9_10

Marentette, J. R., Frank, R. A., Bartlett, A. J., Gillis, P. L., Hewitt, L. M., Peru, K.
M., Headley, J. V., Brunswick, P., Shang, D., & Parrott, J. L. (2015).
Toxicity of naphthenic acid fraction components extracted from fresh and aged oil sands process-affected waters, and commercial naphthenic acid

mixtures, to fathead minnow (Pimephales promelas) embryos. *Aquatic Toxicology*, *164*, 108–117. https://doi.org/10.1016/j.aquatox.2015.04.024

- Marentette, J. R., Sarty, K., Cowie, A. M., Frank, R. A., Hewitt, L. M., Parrott, J. L., & Martyniuk, C. J. (2017). Molecular responses of Walleye (Sander vitreus) embryos to naphthenic acid fraction components extracted from fresh oil sands process-affected water. *Aquatic Toxicology*, *182*, 11–19. https://doi.org/10.1016/j.aquatox.2016.11.003
- McNamara, J. P., & Huber, K. (2018). Metabolic and Endocrine Role of Adipose
 Tissue During Lactation. *Annual Review of Animal Biosciences*, 6, 177–195. https://doi.org/10.1146/annurev-animal-030117-014720
- Muir, L. A., Neeley, C. K., Meyer, K. A., Baker, N. A., Brosius, A. M.,
 Washabaugh, A. R., Varban, O. A., Finks, J. F., Zamarron, B. F., Flesher,
 C. G., Chang, J. S., DelProposto, J. B., Geletka, L., Martinez-Santibanez,
 G., Kaciroti, N., Lumeng, C. N., & O'Rourke, R. W. (2016). Adipose
 tissue fibrosis, hypertrophy, and hyperplasia: Correlations with diabetes in
 human obesity. *Obesity (Silver Spring, Md.)*, 24(3), 597–605.
 https://doi.org/10.1002/oby.21377

<sup>Peng, H., Sun, J., Alharbi, H. A., Jones, P. D., Giesy, J. P., & Wiseman, S. (2016).
Peroxisome Proliferator-Activated Receptor γ is a Sensitive Target for Oil
Sands Process-Affected Water: Effects on Adipogenesis and Identification
of Ligands.</sup> *Environmental Science & Technology*, *50*(14), 7816–7824.
https://doi.org/10.1021/acs.est.6b01890

- Ramírez-Zacarías, J. L., Castro-Muñozledo, F., & Kuri-Harcuch, W. (1992).
 Quantitation of adipose conversion and triglycerides by staining intracytoplasmic lipids with Oil red O. *Histochemistry*, 97(6), 493–497. https://doi.org/10.1007/BF00316069
- Riu, A., Grimaldi, M., le Maire, A., Bey, G., Phillips, K., Boulahtouf, A., Perdu,
 E., Zalko, D., Bourguet, W., & Balaguer, P. (2011). Peroxisome proliferator-activated receptor γ is a target for halogenated analogs of bisphenol A. *Environmental Health Perspectives*, *119*(9), 1227–1232. https://doi.org/10.1289/ehp.1003328
- Rosen, E. D., & MacDougald, O. A. (2006). Adipocyte differentiation from the inside out. *Nature Reviews Molecular Cell Biology*, 7(12), 885–896. https://doi.org/10.1038/nrm2066
- Rosen, E. D., Walkey, C. J., Puigserver, P., & Spiegelman, B. M. (2000). Transcriptional regulation of adipogenesis. *Genes & Development*, 14(11), 1293–1307. https://doi.org/10.1101/gad.14.11.1293
- Scott, A. C., Zubot, W., Davis, C. W., & Brogly, J. (2020). Bioaccumulation potential of naphthenic acids and other ionizable dissolved organics in oil sands process water (OSPW) A review. *Science of The Total Environment*, *712*, 134558.

https://doi.org/10.1016/j.scitotenv.2019.134558

Small, C. C., Cho, S., Hashisho, Z., & Ulrich, A. C. (2015). Emissions from oil sands tailings ponds: Review of tailings pond parameters and emission

estimates. *Journal of Petroleum Science and Engineering*, *127*, 490–501. https://doi.org/10.1016/j.petrol.2014.11.020

- Spiegelman, B. M. (1997). Peroxisome proliferator-activated receptor gamma: A key regulator of adipogenesis and systemic insulin sensitivity. *European Journal of Medical Research*, 2(11), 457–464.
- Tamori, Y., Masugi, J., Nishino, N., & Kasuga, M. (2002). Role of Peroxisome Proliferator-Activated Receptor-γ in Maintenance of the Characteristics of Mature 3T3-L1 Adipocytes. *Diabetes*, 51(7), 2045–2055. https://doi.org/10.2337/diabetes.51.7.2045
- Tontonoz, P., Hu, E., & Spiegelman, B. M. (1994). Stimulation of adipogenesis in fibroblasts by PPAR gamma 2, a lipid-activated transcription factor. *Cell*, 79(7), 1147–1156. https://doi.org/10.1016/0092-8674(94)90006-x
- Vander Meulen, I. J., Schock, D. M., Parrott, J. L., Mundy, L. J., Pauli, B. D., Peru, K. M., McMartin, D. W., & Headley, J. V. (2021). Characterization of naphthenic acid fraction compounds in water from Athabasca oil sands wetlands by Orbitrap high-resolution mass spectrometry. *Science of The Total Environment*, 780, 146342.

https://doi.org/10.1016/j.scitotenv.2021.146342

Vroegrijk, I. O. C. M., van Klinken, J. B., van Diepen, J. A., van den Berg, S. A.
A., Febbraio, M., Steinbusch, L. K. M., Glatz, J. F. C., Havekes, L. M.,
Voshol, P. J., Rensen, P. C. N., van Dijk, K. W., & van Harmelen, V.
(2013). CD36 is important for adipocyte recruitment and affects lipolysis.

Obesity (Silver Spring, Md.), 21(10), 2037–2045.

https://doi.org/10.1002/oby.20354

- Wensveen, F. M., Valentić, S., Šestan, M., Turk Wensveen, T., & Polić, B. (2015). Interactions between adipose tissue and the immune system in health and malnutrition. *Seminars in Immunology*, 27(5), 322–333. https://doi.org/10.1016/j.smim.2015.10.006
- Wiseman, S. B., Anderson, J. C., Liber, K., & Giesy, J. P. (2013). Endocrine disruption and oxidative stress in larvae of Chironomus dilutus following short-term exposure to fresh or aged oil sands process-affected water. *Aquatic Toxicology (Amsterdam, Netherlands)*, 142–143, 414–421. https://doi.org/10.1016/j.aquatox.2013.09.003
- Wu, C., De Visscher, A., & Gates, I. D. (2019). On naphthenic acids removal from crude oil and oil sands process-affected water. *Fuel*, 253, 1229– 1246. https://doi.org/10.1016/j.fuel.2019.05.091
- Young, R. A. (1976). Fat, Energy and Mammalian Survival. *American Zoologist*, *16*(4), 699–710. https://doi.org/10.1093/icb/16.4.699
- Zebisch, K., Voigt, V., Wabitsch, M., & Brandsch, M. (2012). Protocol for effective differentiation of 3T3-L1 cells to adipocytes. *Analytical Biochemistry*, 425(1), 88–90. https://doi.org/10.1016/j.ab.2012.03.005
- Zhang, B., Berger, J., Hu, E., Szalkowski, D., White-Carrington, S., Spiegelman,B. M., & Moller, D. E. (1996). Negative regulation of peroxisome proliferator-activated receptor-gamma gene expression contributes to the

antiadipogenic effects of tumor necrosis factor-alpha. Molecular

Endocrinology, 10(11), 1457–1466.

https://doi.org/10.1210/mend.10.11.8923470

Zwick, R. K., Rudolph, M. C., Shook, B. A., Holtrup, B., Roth, E., Lei, V., Van Keymeulen, A., Seewaldt, V., Kwei, S., Wysolmerski, J., Rodeheffer, M. S., & Horsley, V. (2018). Adipocyte hypertrophy and lipid dynamics underlie mammary gland remodeling after lactation. *Nature Communications*, 9(1), 3592. https://doi.org/10.1038/s41467-018-05911-0

4. Chapter 4: Discussion

There is concern that chemicals released into the environment from oil sands mining activity in the AOSR are impacting the health of exposed wildlife and is contributing to dwindling population of indigenous species (Bakx & Normand, 2018). It has been observed that the baculums of male Lontra canadensis are weaker in animals from areas with high amounts of oil sands activity than areas in the AOSR that are unaffected by oil sands activity(Thomas et al., 2021). This observation may be a mechanism contributing to the lower otter fertility (Gutgesell et al., 2019; Raez-Villanueva et al., 2019) because adequate baculum strength is required for successful reproduction (Stockley et al., 2013). The production and storage of OSPW required by bitumen extraction in the AOSR is one of the primary sources of PRCs in the environment and the toxicity of OSPW is considered to be primarily due to polar organic constituents, including NAFCs (Bartlett et al., 2017). Adipocytes and osteoblasts are two different MSC derived cells that exist in bone, the focus of this thesis is on the effects of NAFCs from OSPW on path selection during differentiation of MSCderived progenitor cells toward osteoblasts generating cortical bone tissue, or adipocytes contributing to BMAT, two important contributors to bone quality and strength.

Bone adapts to changes in loading and repairs microdamage throughout life through the process of remodeling, a tightly regulated process that involves resorption of old bone tissue and the formation of new cortex material. Osteoblasts deposit new cortex material as they differentiate before terminally differentiating into osteocytes or undergoing apoptosis. Our finding that exposure to NAFC inhibits the deposition of ECM at doses of 12.5mg/L and higher provides evidence that NAFC can negatively affect bone formation. We observed increased expression of markers of GR activity with NAFC exposure; however, future experiments treating 3T3-E1 pre-osteoblast cells with a GR antagonist prior to NAFC exposure would be needed to determine whether GR activity is required for the inhibition of osteoblastogenesis caused by NAFC. We also observed that NAFC lowered ECM production to a much greater extent than the GR agonist dexamethasone and is therefore likely inhibiting osteoblast differentiation through other pathways in addition to GR. Although not investigated in our study, NAFCs are well known to induce any hydrocarbon receptor (AHR) activity (Marentette et al., 2017), which is primarily responsible for initiating xenobiotic metabolism in response to environmental chemicals. However, AHR receptor activity has also been shown to induce a pro-aging phenotype in many tissues and impair mitochondrial function (Brinkmann et al., 2020). The deposition of ECM during bone formation is a very energy intensive process. If NAFC exposure impairs mitochondrial function in osteoblasts this could be a plausible mechanism that explains, in part, the inhibition in ECM deposition following NAFC exposure. There are several other pathways by which NAFCs may inhibit osteoblast differentiation including Wnt/Notch, PTH and bone morphogenic proteins (Rutkovskiy et al., 2016). Given the complexity and number of factors

influencing osteoblast differentiation, it may be worthwhile to use -omics techniques to capture a global view of the transcriptional response of osteoblasts to NAFC exposure.

The observation that NAFC inhibit osteoblast differentiation and ECM deposition support our main hypothesis that NAFCs are altering MSC progenitor cell differentiation to impair bone formation. The second avenue by which NAFCs may alter MSC progenitor cell differentiation that we investigated is whether NAFC exposure drives progenitor cells toward a pro-adipogenic phenotype and therefore promotes bone marrow adipogenesis.

Bone marrow adipogenesis involves proliferation of adipocytes in the medullary cavity of long bones and is associated with lower bone strength (Rosen & Bouxsein, 2006). The metabolic function of bone marrow adipose tissue is unknown, however, increases in marrow adiposity are observed in a variety of metabolic diseases and aging (Sebo et al., 2019). In mice, the PPAR γ activating drug rosiglitazone induces increased bone marrow adipogenesis (Lu et al., 2016), which given previous reports that OSPW contains PPAR γ ligands (Kassotis et al., 2018; Peng et al., 2016) led to the hypothesis that components of OSPW may induce marrow adipogenesis. However, we subsequently found that while commercial NA mixtures do contain PPAR γ ligands, NAFCs did not induce adipogenesis. These data did not support our primary hypothesis that NAFC shifted MSC progenitor cell differentiation away from an osteoblast phenotype towards an adipose phenotype. While NAFC exposure did inhibit osteoblast differentiation, it did not induce adipocyte differentiation. In this same study where we investigated whether NAFC induced adipogenesis we also investigated the effects of NAFC in mature adipocytes. We found that the binding of components in NAFCs to PPAR γ are likely antagonistic and inhibit the expression of enzymes required for the regulating lipolysis and the insulin response in adipocytes. These effects are not isolated to bone marrow adipocytes and are likely unrelated to the impact that NAFCs may have on bone strength. However, they do suggest that NAFCs may impair adipose tissue function in exposed animals.

The work presented in this thesis is the first to describe the direct effects of NAFCs on bone cells and pathways affecting osteoblast differentiation, and the first to assess the effects of isolated NAFCs on adipogenesis. In mammals where the baculum is required for successful reproduction, the inhibition of remodeling or formation of the baculum bone by NAFCs or other PRCs would have negative consequences for the reproductive health of these animals. Further, as we have shown that NAFCs can affect glucocorticoid signaling which is involved in numerous physiological process that are essential for life. Therefore, exposure to NAFCs may have a greater impact on the health of mammals than previously thought. This has broad implications for the health of ecosystems in areas affected by oil sands activity like the AOSR.

4.1. Limitations and Future directions

The data presented in this thesis demonstrate potential toxicologic effects of NAFCs on bone and adipose tissue development and metabolism. These experiments were *in vitro* and future experiments investigating whether the observed effects on osteoblasts and adipocytes translate to lower bone strength and impaired bone remodeling in adult animals, and impaired adipose tissue metabolism in vivo. Future studies should investigate the effects of NAFCs on bone formation, and in particular baculum formation, in vivo, as osteoblast differentiation is necessary for skeletal development (Rutkovskiy et al., 2016). The concentrations of NAFCs used in these experiments was based on the concentrations reported in environmental samples, however it is unknown what the concentration of NAFCs are in exposed animals. This is a limitation for the study of the toxicity of NAFC in general, which is complicated by the highly complex nature of NAFC mixtures and the lack of analytical techniques presently available to robustly measure NAFC concentrations in tissue samples (Brown & Ulrich, 2015). Thus, all studies of the toxicity of naphthenic acids in animals will need to estimated exposures concentrations based on levels measured in bodies of water in the environment as we have done in the work presented in this thesis. As analytical techniques are developed to be able to identify tissue concentrations of NAFCs in wildlife, toxicology and pharmacokinetic studies of the effects of NAFCs should be revisited to get a more accurate understanding of what the effects of real-world concentrations of NAFCs are on cells and animals. Another

limitation of the work presented in this thesis is that we observed an induction of markers of GR activity with NAFC treatment, however, we did not determine that these were the sole mechanisms by which NAFC inhibits osteoblast differentiation and ECM deposition. Future studies could use a knock-down of the GR gene or treatment with a selective GR antagonist like RU-43044 prior to treatment with NAFC and see whether the observed effect on ECM deposition is attenuated (Clark, 2008).

In conclusion, the work presented in this thesis presents a mechanism by which NAFCs may impair bone strength and metabolism in exposed animals. Our findings support the hypothesis that exposure to NAFCs alter the differentiation of MSC progenitor cells to favour reduced bone formation. The data presented in this thesis provides evidence that; 1. NAFCs inhibit the differentiation and deposition of ECM of osteoblasts and act, in part, through the GR and 2. NAFCs do not induce adipogenesis of adipocyte precursor cells but rather inhibit PPARγ target gene expression suggesting NAFCs may impair key adipocyte functions in whole body energy homeostasis. Furthermore, this evidence can direct future *in vivo* experiments investigating the effects of NAFC on both bone strength and adipose tissue metabolism, both of which are fundamental to the survival of wildlife.

Bibliography

Akune, T., Ohba, S., Kamekura, S., Yamaguchi, M., Chung, U.-I., Kubota, N., Terauchi, Y., Harada, Y., Azuma, Y., Nakamura, K., Kadowaki, T., & Kawaguchi, H. (2004). PPARgamma insufficiency enhances osteogenesis through osteoblast formation from bone marrow progenitors. *The Journal of Clinical Investigation*, *113*(6), 846–855. https://doi.org/10.1172/JCI19900

Ali, A. A., Weinstein, R. S., Stewart, S. A., Parfitt, A. M., Manolagas, S. C., & Jilka, R. L. (2005). Rosiglitazone causes bone loss in mice by suppressing osteoblast differentiation and bone formation. *Endocrinology*, 146(3), 1226–1235. https://doi.org/10.1210/en.2004-0735

- Antoniadou, E., & David, A. L. (2016). Placental stem cells. Best Practice & Research. Clinical Obstetrics & Gynaecology, 31, 13–29. https://doi.org/10.1016/j.bpobgyn.2015.08.014
- Atkins, G. J., & Findlay, D. M. (2012). Osteocyte regulation of bone mineral: A little give and take. Osteoporosis International: A Journal Established as Result of Cooperation between the European Foundation for Osteoporosis and the National Osteoporosis Foundation of the USA, 23(8), 2067–2079. https://doi.org/10.1007/s00198-012-1915-z
- Bakx, K., & Normand, G. (2018, December 25). Why Indigenous groups are tired of fighting oilsands development.

https://newsinteractives.cbc.ca/longform/drawing-a-line-in-the-oilsandsfight

Barding EE, L. M. (2014). Demographic and Reproductive Characteristics of Reintroduced Northern River Otters in Kentucky: Implications for Population Growth. *American Midland Naturalist*. https://doi.org/172(2):338-347

Bartlett, A. J., Frank, R. A., Gillis, P. L., Parrott, J. L., Marentette, J. R., Brown,
L. R., Hooey, T., Vanderveen, R., McInnis, R., Brunswick, P., Shang, D.,
Headley, J. V., Peru, K. M., & Hewitt, L. M. (2017). Toxicity of
naphthenic acids to invertebrates: Extracts from oil sands process-affected
water versus commercial mixtures. *Environmental Pollution (Barking, Essex: 1987)*, 227, 271–279. https://doi.org/10.1016/j.envpol.2017.04.056

Bateman, M. E., Strong, A. L., McLachlan, J. A., Burow, M. E., & Bunnell, B. A. (2017). The Effects of Endocrine Disruptors on Adipogenesis and Osteogenesis in Mesenchymal Stem Cells: A Review. *Frontiers in Endocrinology*, 7, 171. https://doi.org/10.3389/fendo.2016.00171

Bauer, A. E., Frank, R. A., Headley, J. V., Milestone, C. B., Batchelor, S., Peru, K. M., Rudy, M. D., Barrett, S. E., Vanderveen, R., Dixon, D. G., & Hewitt, L. M. (2019). A preparative method for the isolation and fractionation of dissolved organic acids from bitumen-influenced waters. *Science of The Total Environment*, 671, 587–597. https://doi.org/10.1016/j.scitotenv.2019.03.244
- Boucher, J. G., Boudreau, A., & Atlas, E. (2014). Bisphenol A induces
 differentiation of human preadipocytes in the absence of glucocorticoid
 and is inhibited by an estrogen-receptor antagonist. *Nutrition & Diabetes*,
 4, e102. https://doi.org/10.1038/nutd.2013.43
- Brassey, C. A., Gardiner, J. D., & Kitchener, A. C. (2018). Testing hypotheses for the function of the carnivoran baculum using finite-element analysis. *Proceedings of the Royal Society B: Biological Sciences*, 285(1887).
 https://doi.org/10.1098/rspb.2018.1473
- Bredella, M. A., Torriani, M., Ghomi, R. H., Thomas, B. J., Brick, D. J.,
 Gerweck, A. V., Rosen, C. J., Klibanski, A., & Miller, K. K. (2011).
 Vertebral bone marrow fat is positively associated with visceral fat and
 inversely associated with IGF-1 in obese women. *Obesity (Silver Spring, Md.*), 19(1), 49–53. https://doi.org/10.1038/oby.2010.106
- Brinkmann, V., Ale-Agha, N., Haendeler, J., & Ventura, N. (2020). The Aryl Hydrocarbon Receptor (AhR) in the Aging Process: Another Puzzling Role for This Highly Conserved Transcription Factor. *Frontiers in Physiology*, 10, 1561. https://doi.org/10.3389/fphys.2019.01561
- Briot, K., & Roux, C. (2015). Glucocorticoid-induced osteoporosis. *RMD Open*, *1*(1). https://doi.org/10.1136/rmdopen-2014-000014
- Brown, L. D., & Ulrich, A. C. (2015). Oil sands naphthenic acids: A review of properties, measurement, and treatment. *Chemosphere*, 127, 276–290. https://doi.org/10.1016/j.chemosphere.2015.02.003

- Brunswick, P., Hewitt, L. M., Frank, R. A., Kim, M., van Aggelen, G., & Shang,
 D. (2017). A traceable reference for direct comparative assessment of total naphthenic acid concentrations in commercial and acid extractable organic mixtures derived from oil sands process water. *Journal of Environmental Science and Health, Part A*, *52*(3), 274–280.
 https://doi.org/10.1080/10934529.2016.1253399
- Bukowska, J., Frazier, T., Smith, S., Brown, T., Bender, R., McCarthy, M., Wu,
 X., Bunnell, B. A., & Gimble, J. M. (2018). Bone Marrow Adipocyte
 Developmental Origin and Biology. *Current Osteoporosis Reports*, *16*(3),
 312–319. https://doi.org/10.1007/s11914-018-0442-z
- Canada, N. R. (2016, July 5). *Oil Sands: Water Management*. https://www.nrcan.gc.ca/energy/publications/18750
- Canalis, E. (2005). Mechanisms of glucocorticoid action in bone. *Current* Osteoporosis Reports, 3(3), 98–102. https://doi.org/10.1007/s11914-005-0017-7
- Canalis, E., & Delany, A. M. (2002). Mechanisms of glucocorticoid action in bone. Annals of the New York Academy of Sciences, 966, 73–81. https://doi.org/10.1111/j.1749-6632.2002.tb04204.x
- Canalis, E., Mazziotti, G., Giustina, A., & Bilezikian, J. P. (2007).
 Glucocorticoid-induced osteoporosis: Pathophysiology and therapy.
 Osteoporosis International: A Journal Established as Result of
 Cooperation between the European Foundation for Osteoporosis and the

National Osteoporosis Foundation of the USA, *18*(10), 1319–1328. https://doi.org/10.1007/s00198-007-0394-0

- Casteilla, L., Planat-Bénard, V., Cousin, B., Silvestre, J. S., Laharrague, P.,
 Charrière, G., Carrière, A., & Pénicaud, L. (2005). Plasticity of adipose
 tissue: A promising therapeutic avenue in the treatment of cardiovascular
 and blood diseases? *Archives Des Maladies Du Coeur Et Des Vaisseaux*,
 98(9), 922–926.
- Cawthorn, W. P., Scheller, E. L., Learman, B. S., Parlee, S. D., Simon, B. R., Mori, H., Ning, X., Bree, A. J., Schell, B., Broome, D. T., Soliman, S. S., DelProposto, J. L., Lumeng, C. N., Mitra, A., Pandit, S. V., Gallagher, K. A., Miller, J. D., Krishnan, V., Hui, S. K., ... MacDougald, O. A. (2014). Bone Marrow Adipose Tissue Is an Endocrine Organ that Contributes to Increased Circulating Adiponectin during Caloric Restriction. *Cell Metabolism*, 20(2), 368–375. https://doi.org/10.1016/j.cmet.2014.06.003
- Chen, H., Senda, T., & Kubo, K. (2015). The osteocyte plays multiple roles in bone remodeling and mineral homeostasis. *Medical Molecular Morphology*, 48(2), 61–68. https://doi.org/10.1007/s00795-015-0099-y
- Chen, Q., Shou, P., Zheng, C., Jiang, M., Cao, G., Yang, Q., Cao, J., Xie, N.,
 Velletri, T., Zhang, X., Xu, C., Zhang, L., Yang, H., Hou, J., Wang, Y., &
 Shi, Y. (2016). Fate decision of mesenchymal stem cells: Adipocytes or
 osteoblasts? *Cell Death and Differentiation*, 23(7), 1128–1139.
 https://doi.org/10.1038/cdd.2015.168

- Chen, S.-M., Peng, Y.-J., Wang, C.-C., Su, S.-L., Salter, D. M., & Lee, H.-S. (2018). Dexamethasone Down-regulates Osteocalcin in Bone Cells through Leptin Pathway. *International Journal of Medical Sciences*, 15(5), 507–516. https://doi.org/10.7150/ijms.21881
- Chilelli, M., Griffith, B., & Harrison, D. J. (1996). Interstate Comparisons of River Otter Harvest Data. Wildlife Society Bulletin (1973-2006), 24(2), 238–246. JSTOR.
- Cho, E.-S., Kim, M.-K., Son, Y.-O., Lee, K.-S., Park, S.-M., & Lee, J.-C. (2012).
 The Effects of Rosiglitazone on Osteoblastic Differentiation, Osteoclast
 Formation and Bone Resorption. *Molecules and Cells*, *33*(2), 173–181.
 https://doi.org/10.1007/s10059-012-2240-z
- Cho, S. W., Yang, J.-Y., Her, S. J., Choi, H. J., Jung, J. Y., Sun, H. J., An, J. H.,
 Cho, H. Y., Kim, S. W., Park, K. S., Kim, S. Y., Baek, W.-Y., Kim, J.-E.,
 Yim, M., & Shin, C. S. (2011). Osteoblast-targeted overexpression of
 PPARγ inhibited bone mass gain in male mice and accelerated
 ovariectomy-induced bone loss in female mice. *Journal of Bone and Mineral Research: The Official Journal of the American Society for Bone*and Mineral Research, 26(8), 1939–1952.
 https://doi.org/10.1002/jbmr.366
- Clark, R. D. (2008). Glucocorticoid receptor antagonists. *Current Topics in Medicinal Chemistry*, 8(9), 813–838.

https://doi.org/10.2174/156802608784535011

- Clarke, B. (2008). Normal Bone Anatomy and Physiology. *Clinical Journal of the American Society of Nephrology : CJASN*, *3*(Suppl 3), S131–S139. https://doi.org/10.2215/CJN.04151206
- Cumberland, R. E., & DeVINK, J.-M. (2017). Demographic and Reproductive Characteristics of New Brunswick River Otter Populations Based on 13 Years of Harvest Data. *CWBM*, 6(1). http://cwbm.name/demographic-andreproductive-characteristics-of-new-brunswick-river-otter-populationsbased-on-13-years-of-harvest-data/
- Dallas, S. L., Prideaux, M., & Bonewald, L. F. (2013). The osteocyte: An endocrine cell ... and more. *Endocrine Reviews*, 34(5), 658–690. https://doi.org/10.1210/er.2012-1026
- Danion, M., Deschamps, M.-H., Thomas-Guyon, H., Bado-Nilles, A., Le Floch,
 S., Quentel, C., & Sire, J.-Y. (2011). Effect of an experimental oil spill on
 vertebral bone tissue quality in European sea bass (Dicentrarchus labrax
 L.). *Ecotoxicology and Environmental Safety*, 74(7), 1888–1895.
 https://doi.org/10.1016/j.ecoenv.2011.07.027
- Diamanti-Kandarakis, E., Bourguignon, J.-P., Giudice, L. C., Hauser, R., Prins, G.
 S., Soto, A. M., Zoeller, R. T., & Gore, A. C. (2009). EndocrineDisrupting Chemicals: An Endocrine Society Scientific Statement. *Endocrine Reviews*, *30*(4), 293–342. https://doi.org/10.1210/er.2009-0002
- Doluweera, G., Kralovic, P., & Millington, D. (2017). *Economic Impacts of Canadian Oil and Gas Supply in Canada and the US (2017-2027)* (No.

166; pp. 1–102). CERI. https://ceri.ca/studies/economic-impacts-ofcanadian-oil-and-gas-supply-in-canada-and-the-us-2017-2027

Fang, M., Webster, T. F., Ferguson, P. L., & Stapleton, H. M. (2015).
Characterizing the peroxisome proliferator-activated receptor (PPARγ)
ligand binding potential of several major flame retardants, their
metabolites, and chemical mixtures in house dust. *Environmental Health Perspectives*, *123*(2), 166–172. https://doi.org/10.1289/ehp.1408522

Feige, J. N., Gelman, L., Rossi, D., Zoete, V., Métivier, R., Tudor, C., Anghel, S. I., Grosdidier, A., Lathion, C., Engelborghs, Y., Michielin, O., Wahli, W., & Desvergne, B. (2007). The endocrine disruptor monoethyl-hexyl-phthalate is a selective peroxisome proliferator-activated receptor gamma modulator that promotes adipogenesis. *The Journal of Biological Chemistry*, 282(26), 19152–19166. https://doi.org/10.1074/jbc.M702724200

Fernie, K. J., Marteinson, S. C., Chen, D., Eng, A., Harner, T., Smits, J. E. G., & Soos, C. (2018). Elevated exposure, uptake and accumulation of polycyclic aromatic hydrocarbons by nestling tree swallows (Tachycineta bicolor) through multiple exposure routes in active mining-related areas of the Athabasca oil sands region. *Science of The Total Environment*, 624, 250–261. https://doi.org/10.1016/j.scitotenv.2017.12.123 Friedenstein, A. J., Gorskaja, J. F., & Kulagina, N. N. (1976). Fibroblast precursors in normal and irradiated mouse hematopoietic organs. *Experimental Hematology*, 4(5), 267–274.

General (US), O. of the S. (2004). *The Basics of Bone in Health and Disease*. Office of the Surgeon General (US). https://www.ncbi.nlm.nih.gov/books/NBK45504/

Ghali, O., Broux, O., Falgayrac, G., Haren, N., van Leeuwen, J. P., Penel, G.,
Hardouin, P., & Chauveau, C. (2015). Dexamethasone in osteogenic
medium strongly induces adipocyte differentiation of mouse bone marrow
stromal cells and increases osteoblast differentiation. *BMC Cell Biology*, *16.* https://doi.org/10.1186/s12860-015-0056-6

- Glimcher, M. J. (2006). Bone: Nature of the Calcium Phosphate Crystals and Cellular, Structural, and Physical Chemical Mechanisms in Their Formation. *Reviews in Mineralogy and Geochemistry*, 64(1), 223–282. https://doi.org/10.2138/rmg.2006.64.8
- Government of New Brunswick, C. (2016, October 14). Furbearer Harvest Reports.

https://www2.gnb.ca/content/gnb/en/departments/erd/natural_resources/co ntent/wildlife/content/furbearer_harvest_reports.html

Gutgesell, R. M., Raez-Villanueva, S., Cormier, J., Holloway, A. C., & Thomas,P. J. (2019, December). Trace elements and reproductive success of river otters (Lontra canadensis) in the Alberta Oil Sands Region. *Poster*

Presentation. 51st Annual Society for Toxicology Canada Symposium, Ottawa, Ontario, Canada.

Harada, S., Hiromori, Y., Nakamura, S., Kawahara, K., Fukakusa, S., Maruno, T., Noda, M., Uchiyama, S., Fukui, K., Nishikawa, J., Nagase, H., Kobayashi, Y., Yoshida, T., Ohkubo, T., & Nakanishi, T. (2015). Structural basis for PPARγ transactivation by endocrine-disrupting organotin compounds. *Scientific Reports*, *5*(1), 8520. https://doi.org/10.1038/srep08520

Hasan, A. U., Ohmori, K., Hashimoto, T., Kamitori, K., Yamaguchi, F., Rahman,
A., Tokuda, M., & Kobori, H. (2018). PPARγ activation mitigates
glucocorticoid receptor-induced excessive lipolysis in adipocytes via
homeostatic crosstalk. *Journal of Cellular Biochemistry*, *119*(6), 4627–
4635. https://doi.org/10.1002/jcb.26631

Hauge, E. M., Qvesel, D., Eriksen, E. F., Mosekilde, L., & Melsen, F. (2001).
Cancellous bone remodeling occurs in specialized compartments lined by cells expressing osteoblastic markers. *Journal of Bone and Mineral Research: The Official Journal of the American Society for Bone and Mineral Research*, *16*(9), 1575–1582.
https://doi.org/10.1359/jbmr.2001.16.9.1575

He, Y., Patterson, S., Wang, N., Hecker, M., Martin, J. W., El-Din, M. G., Giesy,J. P., & Wiseman, S. B. (2012). Toxicity of untreated and ozone-treated oil sands process-affected water (OSPW) to early life stages of the fathead

minnow (Pimephales promelas). *Water Research*, *46*(19), 6359–6368. https://doi.org/10.1016/j.watres.2012.09.004

Headley, J. V., & McMartin, D. W. (2004). A review of the occurrence and fate of naphthenic acids in aquatic environments. *Journal of Environmental Science and Health. Part A, Toxic/Hazardous Substances & Environmental Engineering*, *39*(8), 1989–2010.
https://doi.org/10.1081/ese-120039370

Henneicke, H., Gasparini, S. J., Brennan-Speranza, T. C., Zhou, H., & Seibel, M.
J. (2014). Glucocorticoids and bone: Local effects and systemic implications. *Trends in Endocrinology & Metabolism*, 25(4), 197–211. https://doi.org/10.1016/j.tem.2013.12.006

Hewitt, L. M., Roy, J. W., Rowland, S. J., Bickerton, G., DeSilva, A., Headley, J. V., Milestone, C. B., Scarlett, A. G., Brown, S., Spencer, C., West, C. E., Peru, K. M., Grapentine, L., Ahad, J. M. E., Pakdel, H., & Frank, R. A. (2020). Advances in Distinguishing Groundwater Influenced by Oil Sands Process-Affected Water (OSPW) from Natural Bitumen-Influenced Groundwaters. *Environmental Science & Technology*, *54*(3), 1522–1532. https://doi.org/10.1021/acs.est.9b05040

Howard, G. A., Bottemiller, B. L., Turner, R. T., Rader, J. I., & Baylink, D. J.(1981). Parathyroid hormone stimulates bone formation and resorption in organ culture: Evidence for a coupling mechanism. *Proceedings of the*

National Academy of Sciences of the United States of America, 78(5), 3204–3208. https://doi.org/10.1073/pnas.78.5.3204

Huang, R., McPhedran, K. N., Yang, L., & El-Din, M. G. (2016).
Characterization and distribution of metal and nonmetal elements in the Alberta oil sands region of Canada. *Chemosphere*, *147*, 218–229. https://doi.org/10.1016/j.chemosphere.2015.12.099

Janani, C., & Ranjitha Kumari, B. D. (2015). PPAR gamma gene—A review. *Diabetes & Metabolic Syndrome*, 9(1), 46–50. https://doi.org/10.1016/j.dsx.2014.09.015

Kassel, O., & Herrlich, P. (2007). Crosstalk between the glucocorticoid receptor and other transcription factors: Molecular aspects. *Molecular and Cellular Endocrinology*, 275(1–2), 13–29.

https://doi.org/10.1016/j.mce.2007.07.003

- Kassotis, C. D., Bromfield, J. J., Klemp, K. C., Meng, C.-X., Wolfe, A., Zoeller, R. T., Balise, V. D., Isiguzo, C. J., Tillitt, D. E., & Nagel, S. C. (2016).
 Adverse Reproductive and Developmental Health Outcomes Following Prenatal Exposure to a Hydraulic Fracturing Chemical Mixture in Female C57Bl/6 Mice. *Endocrinology*, *157*(9), 3469–3481.
 https://doi.org/10.1210/en.2016-1242
- Kassotis, C. D., Nagel, S. C., & Stapleton, H. M. (2018). Unconventional oil and gas chemicals and wastewater-impacted water samples promote adipogenesis via PPARγ-dependent and independent mechanisms in 3T3–

L1 cells. *The Science of the Total Environment*, 640–641, 1601–1610. https://doi.org/10.1016/j.scitotenv.2018.05.030

Kavanagh, R. J., Frank, R. A., Solomon, K. R., & Van Der Kraak, G. (2013).
Reproductive and health assessment of fathead minnows (Pimephales promelas) inhabiting a pond containing oil sands process-affected water. *Aquatic Toxicology (Amsterdam, Netherlands), 130–131*, 201–209.
https://doi.org/10.1016/j.aquatox.2013.01.007

- Kenkre, J., & Bassett, J. (2018). The bone remodelling cycle. Annals of Clinical Biochemistry, 55(3), 308–327. https://doi.org/10.1177/0004563218759371
- Khosla, S., Oursler, M. J., & Monroe, D. G. (2012). Estrogen and the Skeleton. *Trends in Endocrinology and Metabolism: TEM*, 23(11), 576–581.
 https://doi.org/10.1016/j.tem.2012.03.008

Knag, A. C., Verhaegen, S., Ropstad, E., Mayer, I., & Meier, S. (2013). Effects of polar oil related hydrocarbons on steroidogenesis in vitro in H295R cells. *Chemosphere*, 92(1), 106–115.

https://doi.org/10.1016/j.chemosphere.2013.02.046

- Komori, T. (2016). Cell Death in Chondrocytes, Osteoblasts, and Osteocytes. International Journal of Molecular Sciences, 17(12). https://doi.org/10.3390/ijms17122045
- Komori, T. (2018). Runx2, an inducer of osteoblast and chondrocyte differentiation. *Histochemistry and Cell Biology*, 149(4), 313–323. https://doi.org/10.1007/s00418-018-1640-6

- Leclair, L. A., Pohler, L., Wiseman, S. B., He, Y., Arens, C. J., Giesy, J. P.,
 Scully, S., Wagner, B. D., van den Heuvel, M. R., & Hogan, N. S. (2015).
 In vitro assessment of endocrine disrupting potential of naphthenic Acid
 fractions derived from oil sands-influenced water. *Environmental Science*& *Technology*, 49(9), 5743–5752. https://doi.org/10.1021/acs.est.5b00077
- Lee, K., Jessop, H., Suswillo, R., Zaman, G., & Lanyon, L. (2003). Bone adaptation requires oestrogen receptor-α. *Nature*, 424(6947), 389–389. https://doi.org/10.1038/424389a
- Lee, S., Hong, S., Liu, X., Kim, C., Jung, D., Yim, U. H., Shim, W. J., Khim, J. S., Giesy, J. P., & Choi, K. (2017). Endocrine disrupting potential of PAHs and their alkylated analogues associated with oil spills. *Environmental Science. Processes & Impacts*, 19(9), 1117–1125. https://doi.org/10.1039/c7em00125h
- Leong, W. K., Henshall, T. L., Arthur, A., Kremer, K. L., Lewis, M. D., Helps, S. C., Field, J., Hamilton-Bruce, M. A., Warming, S., Manavis, J., Vink, R., Gronthos, S., & Koblar, S. A. (2012). Human adult dental pulp stem cells enhance poststroke functional recovery through non-neural replacement mechanisms. *Stem Cells Translational Medicine*, *1*(3), 177–187. https://doi.org/10.5966/sctm.2011-0039
- Li, C., Fu, L., Stafford, J., Belosevic, M., & Gamal El-Din, M. (2017). The toxicity of oil sands process-affected water (OSPW): A critical review.

Science of The Total Environment, *601–602*, 1785–1802. https://doi.org/10.1016/j.scitotenv.2017.06.024

Lister, A., Nero, V., Farwell, A., Dixon, D. G., & Van Der Kraak, G. (2008).
Reproductive and stress hormone levels in goldfish (Carassius auratus) exposed to oil sands process-affected water. *Aquatic Toxicology* (*Amsterdam, Netherlands*), 87(3), 170–177.
https://doi.org/10.1016/j.aquatox.2008.01.017

Loughery, J. R., Marentette, J. R., Frank, R. A., Hewitt, L. M., Parrott, J. L., & Martyniuk, C. J. (2019). Transcriptome Profiling in Larval Fathead
Minnow Exposed to Commercial Naphthenic Acids and Extracts from Fresh and Aged Oil Sands Process-Affected Water. *Environmental Science & Technology*, *53*(17), 10435–10444.
https://doi.org/10.1021/acs.est.9b01493

- Lu, W., Wang, W., Wang, S., Feng, Y., & Liu, K. (2016). Rosiglitazone Promotes
 Bone Marrow Adipogenesis to Impair Myelopoiesis under Stress. *PloS One*, *11*(2), e0149543. https://doi.org/10.1371/journal.pone.0149543
- Mangelsdorf, D. J., Thummel, C., Beato, M., Herrlich, P., Schütz, G., Umesono,
 K., Blumberg, B., Kastner, P., Mark, M., Chambon, P., & Evans, R. M.
 (1995). The nuclear receptor superfamily: The second decade. *Cell*, 83(6),
 835–839. https://doi.org/10.1016/0092-8674(95)90199-x
- Marentette, J. R., Frank, R. A., Bartlett, A. J., Gillis, P. L., Hewitt, L. M., Peru, K. M., Headley, J. V., Brunswick, P., Shang, D., & Parrott, J. L. (2015).

Toxicity of naphthenic acid fraction components extracted from fresh and aged oil sands process-affected waters, and commercial naphthenic acid mixtures, to fathead minnow (Pimephales promelas) embryos. *Aquatic Toxicology*, *164*, 108–117. https://doi.org/10.1016/j.aquatox.2015.04.024

- Marentette, J. R., Sarty, K., Cowie, A. M., Frank, R. A., Hewitt, L. M., Parrott, J. L., & Martyniuk, C. J. (2017). Molecular responses of Walleye (Sander vitreus) embryos to naphthenic acid fraction components extracted from fresh oil sands process-affected water. *Aquatic Toxicology*, *182*, 11–19. https://doi.org/10.1016/j.aquatox.2016.11.003
- McKee, R. H., North, C. M., Podhasky, P., Charlap, J. H., & Kuhl, A. (2014).
 Toxicological assessment of refined naphthenic acids in a repeated dose/developmental toxicity screening test. *International Journal of Toxicology*, *33*(1 Suppl), 168S-180S.
 https://doi.org/10.1177/1091581813504229
- Mitra, R. (2011). Adverse effects of corticosteroids on bone metabolism: A review. PM & R: The Journal of Injury, Function, and Rehabilitation, 3(5), 466–471; quiz 471. https://doi.org/10.1016/j.pmrj.2011.02.017
- Moerman, E. J., Teng, K., Lipschitz, D. A., & Lecka-Czernik, B. (2004). Aging activates adipogenic and suppresses osteogenic programs in mesenchymal marrow stroma/stem cells: The role of PPAR-gamma2 transcription factor and TGF-beta/BMP signaling pathways. *Aging Cell*, *3*(6), 379–389. https://doi.org/10.1111/j.1474-9728.2004.00127.x

Mohamad, N.-V., Soelaiman, I.-N., & Chin, K.-Y. (2016). A concise review of testosterone and bone health. *Clinical Interventions in Aging*, 11, 1317– 1324. https://doi.org/10.2147/CIA.S115472

Nagel, S. C., Kassotis, C. D., Vandenberg, L. N., Lawrence, B. P., Robert, J., & Balise, V. D. (2020). Developmental exposure to a mixture of unconventional oil and gas chemicals: A review of experimental effects on adult health, behavior, and disease. *Molecular and Cellular Endocrinology*, *513*, 110722. https://doi.org/10.1016/j.mce.2020.110722

Paschoalini, A. L., Savassi, L. A., Arantes, F. P., Rizzo, E., & Bazzoli, N. (2019).
Heavy metals accumulation and endocrine disruption in Prochilodus argenteus from a polluted neotropical river. *Ecotoxicology and Environmental Safety*, 169, 539–550.

https://doi.org/10.1016/j.ecoenv.2018.11.047

- Peng, H., Sun, J., Alharbi, H. A., Jones, P. D., Giesy, J. P., & Wiseman, S. (2016).
 Peroxisome Proliferator-Activated Receptor γ is a Sensitive Target for Oil
 Sands Process-Affected Water: Effects on Adipogenesis and Identification of Ligands. *Environmental Science & Technology*, *50*(14), 7816–7824.
 https://doi.org/10.1021/acs.est.6b01890
- Peters, L. E., MacKinnon, M., Van Meer, T., van den Heuvel, M. R., & Dixon, D.G. (2007). Effects of oil sands process-affected waters and naphthenic acids on yellow perch (Perca flavescens) and Japanese medaka (Orizias

latipes) embryonic development. *Chemosphere*, 67(11), 2177–2183. https://doi.org/10.1016/j.chemosphere.2006.12.034

Pillai, H. K., Fang, M., Beglov, D., Kozakov, D., Vajda, S., Stapleton, H. M., Webster, T. F., & Schlezinger, J. J. (2014). Ligand binding and activation of PPARγ by Firemaster® 550: Effects on adipogenesis and osteogenesis in vitro. *Environmental Health Perspectives*, *122*(11), 1225–1232. https://doi.org/10.1289/ehp.1408111

Pittenger, M. F., Mackay, A. M., Beck, S. C., Jaiswal, R. K., Douglas, R., Mosca, J. D., Moorman, M. A., Simonetti, D. W., Craig, S., & Marshak, D. R. (1999). Multilineage potential of adult human mesenchymal stem cells. *Science (New York, N.Y.)*, 284(5411), 143–147. https://doi.org/10.1126/science.284.5411.143

Raez-Villanueva, S., Thomas, P. J., Cormier, J., & Holloway, A. C. (2019, June).
Reproductive success of river otters (Lontra canadensis) in the Alberta Oil
Sands Region. *Poster Presentation*. 47th Annual Southern Ontario
Reproductive Biology Meeting, Toronto, Ontario, Canada.

Rauch, A., Seitz, S., Baschant, U., Schilling, A. F., Illing, A., Stride, B., Kirilov,
M., Mandic, V., Takacz, A., Schmidt-Ullrich, R., Ostermay, S., Schinke,
T., Spanbroek, R., Zaiss, M. M., Angel, P. E., Lerner, U. H., David, J.-P.,
Reichardt, H. M., Amling, M., ... Tuckermann, J. P. (2010).
Glucocorticoids Suppress Bone Formation by Attenuating Osteoblast

Differentiation via the Monomeric Glucocorticoid Receptor. Cell

Metabolism, 11(6), 517-531. https://doi.org/10.1016/j.cmet.2010.05.005

- Regulator, A. E. (2019). State of Fluid Tailings Management for Mineable Oil, 2018. Calgary, Alberta.
- Rogers, V. V., Liber, K., & MacKinnon, M. D. (2002). Isolation and characterization of naphthenic acids from Athabasca oil sands tailings pond water. *Chemosphere*, 48(5), 519–527. https://doi.org/10.1016/S0045-6535(02)00133-9
- Rosen, C. J., & Bouxsein, M. L. (2006). Mechanisms of Disease: Is osteoporosis the obesity of bone? *Nature Clinical Practice Rheumatology*, 2(1), 35–43. https://doi.org/10.1038/ncprheum0070
- Rundle, K. I., Sharaf, M. S., Stevens, D., Kamunde, C., & van den Heuvel, M. R.
 (2018). Oil Sands Derived Naphthenic Acids Are Oxidative Uncouplers and Impair Electron Transport in Isolated Mitochondria. *Environmental Science & Technology*, 52(18), 10803–10811. https://doi.org/10.1021/acs.est.8b02638
- Rutkovskiy, A., Stensløkken, K.-O., & Vaage, I. J. (2016). Osteoblast
 Differentiation at a Glance. *Medical Science Monitor Basic Research*, 22, 95–106. https://doi.org/10.12659/MSMBR.901142
- Sebo, Z. L., Rendina-Ruedy, E., Ables, G. P., Lindskog, D. M., Rodeheffer, M.S., Fazeli, P. K., & Horowitz, M. C. (2019). Bone Marrow Adiposity:

Basic and Clinical Implications. *Endocrine Reviews*, 40(5), 1187–1206. https://doi.org/10.1210/er.2018-00138

- Sher, L. B., Woitge, H. W., Adams, D. J., Gronowicz, G. A., Krozowski, Z., Harrison, J. R., & Kream, B. E. (2004). Transgenic expression of 11betahydroxysteroid dehydrogenase type 2 in osteoblasts reveals an anabolic role for endogenous glucocorticoids in bone. *Endocrinology*, 145(2), 922– 929. https://doi.org/10.1210/en.2003-0655
- Sims, N. A., & Martin, T. J. (2015). Coupling Signals between the Osteoclast and Osteoblast: How are Messages Transmitted between These Temporary Visitors to the Bone Surface? *Frontiers in Endocrinology*, 6. https://doi.org/10.3389/fendo.2015.00041
- Sirotkin, A. V., & Harrath, A. H. (2017). Influence of oil-related environmental pollutants on female reproduction. *Reproductive Toxicology (Elmsford, N.Y.)*, 71, 142–145. https://doi.org/10.1016/j.reprotox.2017.05.007
- Sonne, C., Dyck, M., Rigét, F. F., Beck Jensen, J.-E., Hyldstrup, L., Letcher, R. J., Gustavson, K., Gilbert, M. T. P., & Dietz, R. (2015). Penile density and globally used chemicals in Canadian and Greenland polar bears. *Environmental Research*, 137, 287–291.

https://doi.org/10.1016/j.envres.2014.12.026

Stechschulte, L. A., & Lecka-Czernik, B. (2017). Reciprocal Regulation of
 PPARγ and RUNX2 Activities in Marrow Mesenchymal Stem Cells: Fine
 Balance between p38 MAPK and Protein Phosphatase 5. *Current*

Molecular Biology Reports, 3(2), 107–113.

https://doi.org/10.1007/s40610-017-0056-8

Stenson, G. (1975). [PhD Thesis]. University of Alberta.

Stockley, P., Ramm, S. A., Sherborne, A. L., Thom, M. D. F., Paterson, S., & Hurst, J. L. (2013). Baculum morphology predicts reproductive success of male house mice under sexual selection. *BMC Biology*, *11*(1), 66. https://doi.org/10.1186/1741-7007-11-66

Suchacki, K. J., Tavares, A. A. S., Mattiucci, D., Scheller, E. L., Papanastasiou,
G., Gray, C., Sinton, M. C., Ramage, L. E., McDougald, W. A., Lovdel,
A., Sulston, R. J., Thomas, B. J., Nicholson, B. M., Drake, A. J., AlcaideCorral, C. J., Said, D., Poloni, A., Cinti, S., Macpherson, G. J., ...
Cawthorn, W. P. (2020). Bone marrow adipose tissue is a unique adipose
subtype with distinct roles in glucose homeostasis. *Nature Communications*, *11*(1), 3097. https://doi.org/10.1038/s41467-020-16878-

- Tatsumi, S., Ishii, K., Amizuka, N., Li, M., Kobayashi, T., Kohno, K., Ito, M., Takeshita, S., & Ikeda, K. (2007). Targeted ablation of osteocytes induces osteoporosis with defective mechanotransduction. *Cell Metabolism*, 5(6), 464–475. https://doi.org/10.1016/j.cmet.2007.05.001
- Thomas, P. J., Newell, E. E., Eccles, K., Holloway, A. C., Idowu, I., Xia, Z., Hassan, E., Tomy, G., & Quenneville, C. (2021). Co-exposures to trace elements and polycyclic aromatic compounds (PACs) impacts North

American river otter (Lontra canadensis) baculum. *Chemosphere*, 265, 128920. https://doi.org/10.1016/j.chemosphere.2020.128920

Vander Meulen, I. J., Klemish, J. L., Peru, K. M., Chen, D. D. Y., Pyle, G. G., & Headley, J. V. (2021). Molecular profiles of naphthenic acid fraction compounds from mine lease wetlands in the Athabasca Oil Sands Region. *Chemosphere*, 272, 129892.

https://doi.org/10.1016/j.chemosphere.2021.129892

Vander Meulen, I. J., Schock, D. M., Parrott, J. L., Mundy, L. J., Pauli, B. D., Peru, K. M., McMartin, D. W., & Headley, J. V. (2021). Characterization of naphthenic acid fraction compounds in water from Athabasca oil sands wetlands by Orbitrap high-resolution mass spectrometry. *Science of The Total Environment*, 780, 146342.

https://doi.org/10.1016/j.scitotenv.2021.146342

- Vanderschueren, D., Vandenput, L., Boonen, S., Lindberg, M. K., Bouillon, R., & Ohlsson, C. (2004). Androgens and bone. *Endocrine Reviews*, 25(3), 389– 425. https://doi.org/10.1210/er.2003-0003
- Wan, Y. (2010). PPARγ in bone homeostasis. *Trends in Endocrinology* & *Metabolism*, 21(12), 722–728. https://doi.org/10.1016/j.tem.2010.08.006
- Wang, H.-S., Hung, S.-C., Peng, S.-T., Huang, C.-C., Wei, H.-M., Guo, Y.-J., Fu, Y.-S., Lai, M.-C., & Chen, C.-C. (2004). Mesenchymal stem cells in the Wharton's jelly of the human umbilical cord. *Stem Cells (Dayton, Ohio)*, 22(7), 1330–1337. https://doi.org/10.1634/stemcells.2004-0013

- Watt, J., & Schlezinger, J. J. (2015). Structurally-diverse, PPARγ-activating environmental toxicants induce adipogenesis and suppress osteogenesis in bone marrow mesenchymal stromal cells. *Toxicology*, 331, 66–77. https://doi.org/10.1016/j.tox.2015.03.006
- Wu, C., De Visscher, A., & Gates, I. D. (2019). On naphthenic acids removal from crude oil and oil sands process-affected water. *Fuel*, 253, 1229– 1246. https://doi.org/10.1016/j.fuel.2019.05.091
- Yang, O., Kim, H. L., Weon, J.-I., & Seo, Y. R. (2015). Endocrine-disrupting Chemicals: Review of Toxicological Mechanisms Using Molecular Pathway Analysis. *Journal of Cancer Prevention*, 20(1), 12–24. https://doi.org/10.15430/JCP.2015.20.1.12
- Zhang, J., Yang, Y., Liu, W., Schlenk, D., & Liu, J. (2019). Glucocorticoid and mineralocorticoid receptors and corticosteroid homeostasis are potential targets for endocrine-disrupting chemicals. *Environment International*, *133*, 105133. https://doi.org/10.1016/j.envint.2019.105133
- Zhao, C., Irie, N., Takada, Y., Shimoda, K., Miyamoto, T., Nishiwaki, T., Suda,
 T., & Matsuo, K. (2006). Bidirectional ephrinB2-EphB4 signaling controls
 bone homeostasis. *Cell Metabolism*, 4(2), 111–121.
 https://doi.org/10.1016/j.cmet.2006.05.012
- Zhou, B. O., Yue, R., Murphy, M. M., Peyer, J., & Morrison, S. J. (2014). Leptin Receptor-expressing mesenchymal stromal cells represent the main source

of bone formed by adult bone marrow. *Cell Stem Cell*, *15*(2), 154–168. https://doi.org/10.1016/j.stem.2014.06.008